





The TREADON Trial Protocol

Full title: Clinical and cost-effectiveness of individualised exercises and foot orthoses in the treatment of plantar heel pain: a randomised multi-arm multi-stage (MAMS) adaptive trial

Short title: TReatments of Exercise AnD Orthotics for plaNtar heel pain: TREADON

RESEARCH REFERENCE NUMBERS:

IRAS Number: 314272 ISRCTN: ISRCTN12418153 Sponsor Number: RG-0344-22 Funder's Number: NIHR131638

PROTOCOL VERSION NUMBER: 2.0 ISSUE DATE: 08-Dec-2022

Funding acknowledgement

This project is funded by the NIHR Health Technology Assessment programme (NIHR131638).

Department of Health and Social Care disclaimer

The views expressed are those of the authors and not necessarily those of the NHS, the NIHR or the Department of Health and Social Care.

SIGNATURE PAGE

For Keele University sponsored studies, the sponsor will confirm approval of the protocol by signing the IRAS form and therefore a signature on the protocol is not required. The sponsor must be notified of all amendments to the protocol, both substantial and non-substantial. Review of amendments by the sponsor will act as the confirmation that the sponsor confirms approval of the amended protocol.

The undersigned confirm that the following protocol has been agreed and accepted and that the Chief Investigator agrees to conduct the research in compliance with the approved protocol, GCP guidelines, the Sponsor's SOPs, and other regulatory requirements as amended.

I agree to ensure that the confidential information contained in this document will not be used for any other purpose other than the evaluation or conduct of the clinical investigation without the prior written consent of the Sponsor.

I also confirm that I will make the findings of the trial publicly available through publication or other dissemination tools without any unnecessary delay and that an honest accurate and transparent account of the trial will be given; and that any discrepancies from the trial as planned in this protocol will be explained.

Chief Investigator:

Signature:

Name (please print):

Professor Edward Roddy

Sponsor statement:

Where Keele University takes on the sponsor role for protocol development oversight, the signing of the IRAS form by the Sponsor will serve as confirmation of approval of this protocol.

Date: 08/12/2022

LIST of CONTENTS

GENERAL INFORMATION	Page No.
TITLE PAGE	1
RESEARCH REFERENCE NUMBERS	1
SIGNATURE PAGE	2
LIST of CONTENTS	3
LIST OF ABBREVIATIONS	4
KEY TRIAL CONTACTS	6
TRIAL SUMMARY	10
PLAIN ENGLISH SUMMARY	12
TRIAL FLOW CHARTS	13
SECTION	
1. BACKGROUND	19
2. RATIONALE	19
3. OBJECTIVES AND OUTCOME MEASURES/ENDPOINTS	21
4. TRIAL DESIGN	22
5. TRIAL SETTING	28
6. ELIGIBILITY CRITERIA	28
7. TRIAL PROCEDURES	29
8. STATISTICS AND DATA ANALYSIS	40
9. DATA HANDLING	43
10. MONITORING & AUDIT	45
11. ETHICAL AND TRIAL ADMINISTRATION	48
12. DISSEMINATION POLICY	53
13. REFERENCES	56
14. APPENDICES	61

LIST OF ABBREVIATIONS

Abbreviation	Description
CEAC	Cost Effectiveness Acceptability Curve
CI	Chief Investigator
CRF	Case Report Form
CRN	Clinical Research Network
СТИ	Clinical Trials Unit
DMC	Data Monitoring Committee
FPI	Foot Posture Index
GCP	Good Clinical Practice
GDPR	General Data Protection Regulation
GP	General Practitioner
IAU	Impact Accelerator Unit
ICER	Incremental Cost-effectiveness ratio
DMC	Data Monitoring Committee
DNA	Did Not Attend
HEAP	Health Economic Analysis Plan
ISF	Investigator Site File
ISRCTN	International Standard Randomised Controlled Trials Number
ІТТ	Intention-to-Treat
MAMS	Multi-Arm, Multi-Stage
MAR	Missing At Random
MCAR	Missing Completely At Random
MDC	Minimum Data Collection
MFPDI	Manchester Foot Pain and Disability Index
MNAR	Missing Not At Random
NHS	National Health Service
NICE	National Institute for Health and Care Excellence
NRS	Numeric Rating Scale
PHP	Plantar Heel Pain
PI	Principal Investigator
PIL	Participant Information Leaflet
PPIE	Patient and Public Involvement and Engagement
PSS	Personal Social Services
QA	Quality Assurance
QALY	Quality Adjusted Life Year
RCT	Randomised Control Trial
REC	Research Ethics Committee

RUG	Research User Group
SAE	Serious Adverse Event
SMA	Self-Management Advice
SOP	Standard Operating Procedure
TMG	Trial Management Group
TSC	Trial Steering Committee
URL	Unique Resource Locator
UTA	Unable To Attend

KEY TRIAL CONTACTS

Chief Investigator	Professor Edward Roddy Professor of Rheumatology and Honorary Consultant Rheumatologist, School of Medicine, Keele University, Staffordshire, ST5 5BG. Tel: 01782 734715; Fax: 01782 734719 Email: e.roddy@keele.ac.uk
Associate / Principal Investigators	Dr Martin Thomas (Keele AI) Senior Research Fellow in Clinical Epidemiology, School of Medicine, Keele University, Staffordshire, ST5 5BG. Tel: 01782 734874; Fax: 01782 734719 Email: m.thomas@keele.ac.uk
	Dr Gordon Hendry (Glasgow PI) Senior Research Fellow, Musculoskeletal Health Research Group, Research Centre for Health, Glasgow Caledonian University, Glasgow, G4 0BA. Tel: 0141 331 3635 Email: Gordon.hendry@gcu.ac.uk
	Professor Anne-Maree Keenan OBE (Leeds PI) Professor Applied Health Research, School of Healthcare, University of Leeds, Leeds, LS2 9JT. Tel: 0113 343 1238. Email: a.keenan@leeds.ac.uk
Sponsors' representative	Head of Project Assurance, Directorate of Research, Innovation and Engagement, Innovation Centre 2 Keele University, Staffordshire ST5 5NH Tel: 01782 732975 research governance@keele.ac.uk
Funder	National Institute for Health and Care Research, Health Technology Assessment Programme (HTA), Evaluation, Trials and Studies Co-ordinating Centre, University of Southampton, Alpha House, Enterprise Road, Southampton, SO16 7NS Tel: 023 8059 57131 Email: netscomms@nihr.ac.uk
Trial Management	Keele Clinical Trials Unit (CTU), David Weatherall Building, Keele University, Staffordshire, ST5 5BG Tel: 01782 732950; Fax: 01782 733911 Ctu.operations@keele.ac.uk
Key Protocol Contributors	Dr Melanie Holden Senior Research Fellow, School of Medicine, Keele University, Staffordshire, ST5 5BG. Tel: 01782 734720; Fax: 01782 734719 Email: m.holden@keele.ac.uk
	Dr Pavel Mozgunov Senior Research Associate, MRC Biostatistics Unit, University of Cambridge, Cambridge, CB2 0SR. Tel: 07493 182881; Fax: 01223 330365 Email: p.mozgunov@mrc-bsu.cam.ac.uk

Professor Thomas Jaki Professor of Statistics and MRC Investigator, MRC Biostatistics Unit, East Forvie Building, University of Cambridge, Cambridge, CB2 0SR. Tel: 07971 663463 Email: thomas.jaki@mrc-bsu.cam.ac.uk
Professor Nadine Foster Director of STARS Education and Research Alliance Conjoint Professor, The University of Queensland, Faculty of Health and Behavioural Sciences and Metro North Health, Visiting Professor of Musculoskeletal Health, School of Medicine, Keele University, Staffordshire, ST5 5BG. Tel: 07336 55229 Email: n.foster@uq.edu.au
Professor Christian Mallen NIHR Research Professor in General Practice, School of Medicine, Keele University, Staffordshire, ST5 5BG. Tel: 01782 734879; Fax: 01782 734719 Email: c.d.mallen@keele.ac.uk
Professor Hylton B Menz NHMRC Senior Research Fellow, Discipline of Podiatry, La Trobe Sport and Exercise Medicine Research Centre, School of Allied Health, Human Service and Sport, La Trobe University, Victoria 3086 Australia, Health Sciences Building 3, Room 540. Tel: +61 3 9479 5801; Fax: +61 3 9479 5415 Email: h.menz@latrobe.edu.au
Dr Jesse Kigozi Assistant Professor in Health Economics, Institute of Applied Health Research, University of Birmingham, Health Economics Unit, IOEM Building, University of Birmingham, Edgbaston, Birmingham, B15 2TT. Tel: 0121 414 8971 Email: j.kigozi@bham.ac.uk
Dr Thomas Burnett Lecturer in Statistics, Department of Mathematical Sciences, University of Bath, 4 West, University of Bath, Claverton Down, Bath, BA2 7AY. Tel: 01225 387694 Email: tb292@bath.ac.uk
Mr Chris Drake NIHR Doctoral Research Fellow, Leeds Institute of Rheumatic and Musculoskeletal Medicine, University of Leeds, Leeds LS7 4SA. Tel: 0113 3924939 Email: c.j.drake@leeds.ac.uk

	Mrs June Brammar
	Member, Patient and Public Involvement and Engagement (PPIE)
	(PPIE) and Research User Group Member, Keele
	University, Staffordshire, ST5 5BG.
	Tel: 01782 734844; Fax: 01782 734719 Email: jupe brammar@moft.phs.uk
Lead Statistician	Dr Martyn Lewis
	Reader in Statistics, School of Medicine, Keele University,
	Staffordshire, ST5 5BG.
	Tel: 01782 734849; Fax: 01782 734719
Trial Steering Committee	Professor Karen Barker
	Professor of Physiotherapy, Nuffield Department of
	Orthopaedics, Rheumatology and Musculoskeletal
	Sciences, University of Oxford, Headington, Oxford, OX3
	7⊓⊏. Email: Karen barker@ndorms ox ac uk
	Professor Catherine Bowen
	Environmental and Life Sciences, Building 67 Rom 4013
	University of Southampton, Southampton, SO17 1BJ.
	Tel: (0)23 8059 7637
	Email: C.J.Bowen@soton.ac.uk
	Dr Helen French
	Senior Lecturer in Physiotherapy, School of Physiotherapy,
	RCSI University of Medicine and Health Sciences, 123 St
	Stephens Green, Dublin 2.
	Email: hfrench@rcsi.com
	Dr Philip Pallman Research Fellow (Statistics), Centre for Trials Research
	University of Cardiff, 505, 5 th Floor Neuadd Meirionnydd,
	University Hospital of Wales, Heath Park, Cardiff, CF14
	4YS.
	Tel: (U)29 2068 7461 Email: pallmannn@cardiff.ac.uk
	ะเกล. หลากลากษะเลเนท.ลง.นห
	Ms Peggysue Evans
	Kesearch User Group member, C/U Adele Higginbottom,
	5BG.
	Ma Ida Pantlay
	Research User Group member C/O Adele Higginbottom
	School of Medicine, Keele University, Staffordshire, ST5
	5BG.
Data Monitoring Committee	Professor Alan Montgomery
	Professor of Medical Statistics and Clinical Trials,
	Nottingham Clinical Trials Unit, University of Nottingham,

University Park, Nottingham, NG7 2RD. Tel: 0115 82 31612 Email: Alan.Montgomery@nottingham.ac.uk
Professor Caroline McIntosh Professor of Podiatric Medicine, Head of School of Health Sciences, National University of Ireland Galway, Ireland, H91 TK33. Email: caroline.mcintosh@nuigalway.ie
Professor Rebecca Kearney Warwick Clinical Trials Unit, Warwick School of Medicine, University of Warwick, Coventry, CV4 7AL. Tel: 02476 573156 Email: R.S.Kearney@warwick.ac.uk

TRIAL SUMMARY

Trial Title	Clinical and cost-effectiveness of individualised exercises and foot orthoses in the treatment of plantar heel pain: a randomised multi-arm multi-stage (MAMS) adaptive trial					
Internal Ref. Number (or short title)	TReatments of Exercise AnD Orthotics for plaNtar heel pain: TREADON					
Trial Design	Interventional Allocation: randomised; Intervention model: four-arm two- stage adaptive parallel-group; Masking: trial administrator, data entry administrator, lead trial statisticians Primary purpose: Symptom control/reduction & reducing functional limitation Phase III					
Trial Intervention (where applicable)	Self-management advice (SMA	A) booklet (control arm) versus				
	SMA booklet plus individualise	d exercises or				
	SMA booklet plus prefabricated	d foot orthoses <i>or</i>				
	SMA booklet plus individualise foot orthoses	d exercises and prefabricated				
Trial Participants	Adults aged 18 years or older with plantar heel pain (PHP)					
Planned Sample Size	Up to 696 (anticipated range based on adaptive trial design: 490-610)					
Treatment duration	12 weeks					
Follow up duration	12 months					
Planned Trial Period	01 January 2022 – 31 March 2026					
	Objectives	Outcome Measures				
Primary	To determine, in adults with PHP, whether a SMA booklet combined with individualised exercises and/or prefabricated foot orthoses leads to greater improvement in pain in the medium term (average pain over 6-12 weeks of follow-up) than a SMA booklet alone.	Change in PHP intensity score 0-10 numeric rating scale (NRS) between baseline and average NRS score during weeks 6-12.				
Secondary	To compare, in adults with PHP, the effect of an SMA booklet combined with individualised exercises and/or prefabricated foot orthoses compared with an SMA booklet alone on: i) short-term pain trajectories over weeks 1 to 12, including individual weekly comparisons	PHP severity score 0-10 numeric rating scale, first step pain on a 0-10 numeric rating scale; Foot Function Index; patient global rating of change; pain self-efficacy questionnaire; brief illness perceptions questionnaire; quality of life (EQ5D-5L); work loss and presenteeism; self-reported healthcare use for plantar bool pain (NHS)				

 ii) pain at 6 and 12 months, and monthly from 3 months to month 12 iii) first step pain, physical function, patient global rating of change, pain self-efficacy, illness perceptions, ability to work, and treatment satisfaction at 12 weeks, 6 and 12 months. 	and private); self-reported treatment adherence; satisfaction with care; treatment credibility; adverse events.
iv) cost-effectiveness of an SMA booklet combined with individualised exercises and/or prefabricated foot orthoses with an SMA booklet alone.	

PLAIN ENGLISH SUMMARY

AIMS OF THE RESEARCH: To show whether exercises and/or foot orthoses (shoe insoles) provide more pain relief for adults with plantar heel pain than a self-management advice booklet alone, and whether this is good value for money for the NHS.

BACKGROUND: Pain under the heel (called plantar heel pain, PHP) is a common condition affecting 1 in 10 adults during their lifetime. It makes walking and completing everyday tasks, including going to work, difficult. Most patients who consult in general practice are given pain medication and advice, yet the problem often continues. In our successful pilot and feasibility trial, foot orthoses and exercises showed promise for improving pain and function, but a large, high-quality main trial is needed to confirm this.

DESIGN AND METHODS: In this randomised controlled trial, up to 696 adults with PHP will be given a self-management advice booklet and allocated by chance to one of four treatments: exercise, foot orthoses, combined exercises and foot orthoses, or no additional treatment. Participants will be sent a weekly text-message to collect pain scores for up to 12 weeks and then monthly from 3 to 12 months, as well as a questionnaire at 3, 6 and 12 months. Half-way through the trial, only treatments that appear to be reducing pain will continue to be offered and the trial will be stopped early if there is strong evidence that one treatment is better than the others, making the trial good value for money.

PATIENT AND PUBLIC INVOLVEMENT AND ENGAGEMENT: Patient representatives were involved in our pilot and feasibility trial and will support every stage of the main trial. Our lay co-investigator helped write this summary. Five patients with PHP took part in a workshop to help design the main trial. For example, they helped us decide how often pain should be measured. We will also involve patients to assist with our recruitment strategy and help us to interpret the trial findings, develop easily understandable messages to explain the findings and publicise the findings widely.

DISSEMINATION: We will present and publish the findings at healthcare conferences, in medical journals, on relevant websites, through social media, and in general practices and physiotherapy and podiatry services. A summary of the results will be provided to participants when the trial is completed. Articles about the trial will also be written for local magazines, newspapers, and radio.













1 BACKGROUND

Plantar heel pain (PHP) is a term describing several undifferentiated painful conditions affecting the plantar heel. It is now preferred to the label "plantar fasciitis", which was used commonly in the past.¹ It is the most prevalent soft tissue foot complaint, affecting 10% of adults, and impairs mobility, foot and physical function, and ability to work, impacting negatively on quality of life.²⁻⁴ Clinical features include pain under the heel, made worse by weight-bearing, particularly after prolonged rest. Its specific cause is uncertain although established risk factors include obesity; pronated foot posture; reduced ankle or first metatarsophalangeal joint range of motion; prolonged weight-bearing; and tightness in the gastrocnemius and soleus muscles, plantar fascia and Achilles tendon.^{2,5} Weakness of the gastrocnemius, soleus, and intrinsic foot muscles has also been implicated.⁶

PHP is commonly thought to be a benign, self-limiting condition that, for most, resolves within 1 year.⁷ However, symptoms can become chronic and persistent: almost half report persistent symptoms after 10 years, leading to impaired quality of life, physical inactivity and weight gain.^{4,8} A NICE Clinical Knowledge Summary recommends initial treatment with analgesia and advice regarding rest, footwear, heel pads, weight loss and stretching exercises.⁷ Referral to a podiatrist or physiotherapist is advised if mild symptoms persist beyond a few months despite conservative treatment or if symptoms are severe. In the Netherlands and Australia, the most frequent strategies employed by General Practitioners (GPs) to manage PHP are referral to a podiatrist (12-20%), watchful waiting (18-37%), NSAIDs (18-28%), and advice to wear insoles (16%).^{9,10} In our population-based observational trial of 5,109 adults in the UK,^{2,11} 80% of people with PHP did not improve significantly over 18 months and only 40% saw a podiatrist or physiotherapist in that time (unpublished data). Qualitative studies show that the expectations and needs of people with PHP are often unmet.¹²

Podiatrists and physiotherapists commonly use foot orthoses and/or lower limb exercises to treat patients with PHP.¹³ Foot orthoses are insole devices designed to optimise foot loading distribution and adjust medial longitudinal arch function through control of specific foot motion.¹⁴ They reduce rearfoot pronation in a dose-dependent manner, addressing planus foot posture (flat feet) and excessive foot pronation (eversion) associated with PHP, and increase foot-to-surface contact area, lowering plantar heel pressures and tensile stresses at the calcaneal-plantar fascia junction during loading to provide symptomatic relief from PHP.¹⁴⁻¹⁸ Exercises, including stretching the Achilles tendon and plantar fascia and strengthening of the intrinsic muscles of the foot and lower limb muscles (e.g. hamstrings, quadriceps and triceps surae), aim to improve movement and restore normal foot loading.¹⁹⁻²¹ Rearfoot eversion and tightness and weakness in the plantar fascia, Achilles tendon and foot muscles are therefore viable therapeutic targets for foot orthoses and individualised exercises respectively in the treatment of PHP. However, orthoses and exercises are infrequently used to treat this common and disabling condition despite many patients having persistent problems.

2 RATIONALE

PHP is the most common musculoskeletal cause of foot pain and is mostly managed in primary care.² It commonly affects people of working age, and leads to pain, reduced mobility, impaired quality of life and difficulties with everyday activities including work.²⁻⁴ People with plantar heel pain commonly feel their problem is ignored and not taken seriously by health professionals.¹²

The evidence-base to inform clinical decisions about treatments is limited and often of poor quality.²²⁻²⁴ Despite the plausibility of foot orthoses and exercises as interventions for PHP, recent systematic reviews highlight limitations of the existing evidence about their effectiveness. Many published randomised controlled trials (RCTs) are limited by small sample sizes, short-term follow-up and poor methodological quality, illustrating the need for larger trials incorporating longer follow-up, more robust methods, and higher reporting standards.²²⁻²⁴ Two systematic reviews of foot orthoses compared with sham orthoses for PHP published in 2018 reached different conclusions despite pooling data from the same three trials: one found moderate-quality evidence that foot orthoses are effective at reducing pain in the medium term (7-12 weeks) but had no effect in the short term or on function,²² whereas the other found that foot orthoses were ineffective.²³ This difference has been attributed to the reviews extracting outcome data for different measures of foot pain.²⁵ Our recent network meta-analysis found

that exercises alone did not reduce pain or improve function in patients with PHP in the short or medium term, but improved function at 12 months compared with placebo/sham interventions based on only two small RCTs (<50 participants each).²⁴ No treatments commonly used for the management of PHP (including orthoses and exercises) were significantly better than any other for pain and function. Placebo/sham interventions and NSAIDs were the least effective, suggesting that current recommendations to manage PHP first-line with analgesics, NSAIDs and watchful waiting may be suboptimal. We found only one RCT which examined the effectiveness of combining orthoses and exercises.²⁶ Prefabricated orthoses plus stretching exercises were more effective than stretching exercises alone although the trial was small (42-51 participants per arm), follow-up was short (8 weeks) and risk of bias was high.

Subsequently, one small RCT (n=95) reported greater improvement in pain with usual podiatry care plus physical therapy than with usual podiatry care alone at 1 year, but there was no difference in physical function.²⁷ A larger RCT (n=185) concluded that referral to a podiatrist for custom-made insoles does not lead to better outcomes compared with usual GP care.²⁸ However, usual GP care within the trial was more intensive than in routine practice with 41% of participants receiving biomechanical interventions such as heel cups and 15% corticosteroid injection.^{10,29} Key differences from our trial are the type of orthosis (prefabricated, not custom-made), shorter follow-up (6 months versus 12 months), and lack of standardisation of the orthosis intervention. We found a small ongoing pilot RCT (48 participants with PHP) which compares radial shockwave therapy plus exercises versus exercises alone and a small ongoing feasibility study (n=20) for a RCT of foot exercises plus education versus brief advice.^{30,31} We did not find any large pragmatic RCTs comparing the clinical and cost effectiveness of a combination of exercises and orthoses against usual care for PHP.

Treatments such as foot orthoses, individualised exercises or a combination of both show promise for improving pain and function in people with PHP and have potential to be cost-effective by reducing further healthcare use and facilitating earlier return to everyday activities including work. There is currently no evidence of cost-effectiveness of exercise programmes for PHP, with evidence for orthoses limited to one study which showed that prefabricated orthoses were similarly effective to customised orthoses but significantly less expensive.³² High-quality evidence to support their clinical and cost-effectiveness for PHP either alone or in combination in a primary care population is needed to guide patients, clinicians and commissioners whether these treatments should be more routinely offered. Informed by our successful feasibility and pilot trial³³, this definitive trial will investigate the clinical and cost-effectiveness of individualised exercises and/or pre-fabricated foot orthoses plus good self-management advice compared with good self-management advice alone for adults with PHP in primary care over the medium and long term. Such a trial is needed to support clinical decision-making and inform NHS policy and commissioning pathways for this patient group who experience considerable pain and functional limitation but are often overlooked in terms of treatment options that are available but only accessed by a few.^{9,10}

2.1 The TREADON pilot and feasibility trial

We completed the TReatments of Exercise AnD Orthotics for plaNtar heel pain (TREADON) pilot and feasibility trial.³³

Eighty-two participants with PHP were randomised to one of four intervention arms:

(1) a self-management advice (SMA) booklet providing advice about exercises, pain relief, footwear, and weight loss (as per NICE Clinical Knowledge Summary⁷)

- (2) SMA booklet and individualised exercises
- (3) SMA booklet and prefabricated foot orthoses
- (4) SMA booklet, individualised exercises and prefabricated foot orthoses.

The trial demonstrated feasibility in terms of participant identification and recruitment methods; intervention fidelity; intervention delivery by both physiotherapists and podiatrists; adherence satisfaction, and credibility of the three clinician-supported interventions; and retention over 12 weeks. Important information to inform the choice of primary outcome and sample size was obtained. We successfully demonstrated the feasibility and acceptability of conducting this large RCT investigating

the costs and long-term clinical effectiveness of individualised exercises and/or prefabricated foot orthoses on pain and function for PHP that will inform clinical practice.

3 OBJECTIVES AND OUTCOME MEASURES/ENDPOINTS

The overall aim of the trial is to compare the additional benefit of individualised exercises and/or prefabricated foot orthoses versus a self-management advice (SMA) booklet alone to treat adults with plantar heel pain (PHP) in primary care.

3.1 Primary objective/Research question

In adults with PHP, does a SMA booklet combined with individualised exercises and/or prefabricated foot orthoses lead to greater improvement in pain in the medium term (average pain over 6-12 weeks of follow-up) than a SMA booklet alone?

3.2 Secondary objectives

To compare, in adults with PHP:

- 3.2.1 The effect of an SMA booklet combined with individualised exercises and/or prefabricated foot orthoses with a SMA booklet alone on;
 - i) short-term pain trajectories over weeks 1 to 12, including individual weekly comparisons,
 - ii) pain at 6 and 12 months, and monthly from 3 months to month 12,
 - iii) first step pain, physical function, patient global rating of change, pain self-efficacy, illness perceptions, ability to work, and treatment satisfaction at 12 weeks, 6 and 12 months.
- 3.2.2 The cost-effectiveness of an SMA booklet combined with individualised exercises and/or prefabricated foot orthoses with an SMA booklet alone.

3.3 Outcome measures/endpoints

To determine the primary outcome measure for the main trial, three outcomes were evaluated as part of the TREADON pilot and feasibility trial.³³ These included:

- i) pain intensity change score using a 0-10 numeric rating scale (NRS),
- ii) change in foot pain measured using the Foot Function Index pain subscale³⁴,
- iii) change in foot pain measured using the Manchester Foot Pain and Disability Index (MFPDI) pain subscale.³⁵

Due to limited responsiveness to change over time, reported in previous studies^{36,37}, the MFPDI was considered unsuitable for the main trial.³³ In the TREADON feasibility and pilot trial, both the pain intensity NRS and Foot Function Index pain subscales completion rates, responsiveness to change, and correlation with other change scores supported their use in a future main trial.³³ The NRS pain score has a more versatile application as part of weekly data collection using SMS text-message and was therefore selected as the primary outcome³³, with the Foot Function Index being retained as an important secondary outcome.

The end points are defined as:

• Primary end point at 6-12 weeks for clinical effectiveness and at 12 months for cost-effectiveness analysis.

3.4 Primary endpoint/outcome

The primary outcome for clinical effectiveness is change in PHP intensity (0-10 NRS) between the baseline and the average rating over weeks 6 to 12. Outcome data are patient-reported and will be collected by SMS text-message (weekly between weeks 1-12, monthly thereafter until 12 months) and questionnaires at 12 weeks and 6 and 12 months.

3.5 Secondary endpoints/outcomes

Secondary outcomes will comprise PHP severity score (0-10 NRS), first step pain (0-10 NRS), Foot Function Index,³⁴ (pain, disability, activity restriction subscales, and overall), patient global rating of change, pain self-efficacy questionnaire,³⁸ brief illness perceptions questionnaire,³⁹ quality of life (EQ5D-5L),⁴⁰ work loss and presenteeism,⁴¹ self-reported healthcare use for PHP (NHS and private), self-reported treatment adherence, treatment credibility, satisfaction with care, and adverse events.

<u>Adverse events:</u> Trial-related adverse events (for example skin irritation from orthoses, muscle soreness from exercise) will be captured through case report forms (CRF) completed by TREADON physiotherapists/podiatrists, and direct contact between the CTU and the participant, their TREADON physiotherapist/podiatrist, GP or site PI. Participants randomised to the clinician-supported intervention arms will record adverse events in a weekly diary for the 12-week intervention period.

Trial safety reporting procedures will ensure any unexpected serious adverse events which are deemed related to the trial will be reported to the Research Ethics Committee and Sponsor. All SAEs either confirmed or suspected to be related to the trial procedures will be reviewed by the Data Monitoring Committee and reported to the Trial Steering Committee.

4 TRIAL DESIGN

Multi-centre, primary care-based, randomised, parallel group, four-arm two-stage adaptive trial, with 6-month internal pilot.

4.1 Interventions/Treatments

The duration of the intervention regardless of group allocation will be 12 weeks. Participants will be asked not to use other types of treatments, other than medication that their GP has provided, during the intervention period if possible; however any additional healthcare and self-care use will be recorded in the 12-week follow-up questionnaire, 6 month follow-up questionnaire and 12 month follow-up questionnaire.

4.1.1 Self-management advice (SMA) booklet (control arm)

Participants randomised to receive SMA only will be mailed a SMA booklet, based on the Versus Arthritis leaflets on plantar fasciitis⁴² and foot and ankle pain,⁴³ supplemented with specific advice and information including a small number of stretching exercises and self-help messages about pain relief, footwear, rest and weight loss. This will be consistent with best practice guidance.⁴⁴ The booklet describes five stretching exercises for the plantar fascia and Achilles tendon to be performed twice per day⁴², without instructions for individualisation, progression or supervision of exercises.

4.1.2 SMA booklet plus individualised exercises (SMA-exercises)

Participants randomised to SMA-exercises will be given the SMA booklet at the initial treatment appointment. The treating clinician (a physiotherapist or podiatrist) will assess foot posture and function to determine the exercise type and dose. At the discretion of the clinician, a more generic lower limb assessment of alignment and function can also be undertaken, enabling strengthening and stretching for hip abductors, quadriceps and hamstrings to be prescribed if deemed to be important as part of the overall PHP treatment. Exercise selection will be informed by the level of clinically-observed muscle tightness, weakness and functional limitation (Figure 7). Exercise dose will be individualised and progressed, based on assessment findings and informed by current exercise guidelines.⁴⁵ The exercises are drawn from best available evidence^{5,19-21,46-47} and discussion with clinicians during workshops prior to our pilot and feasibility trial. Exercises will include foot-specific stretches/exercises targeting the plantar fascia, intrinsic foot muscles, Achilles tendon, key ankle related muscle groups such as soleus and gastrocnemius, and other muscle groups in the lower limb identified as targets within the assessment. Participants will be taught how to perform and progress these exercises and

given an individualised and detailed exercise sheet (online or paper) describing the regimen and showing photographs of the exercises. The exercise sheets provided will be trial-specific exercises selected by the clinician and comprising photographs and written instruction completed by the clinician. Participants will be followed up for up to 6 treatment appointments over 12 weeks at the discretion of the treating clinician (face-to-face appointments prioritised if possible, or virtual, telephone, if required). We will aim for appointment lengths of 1 hour for the initial appointment and 30 minutes for follow-up appointments.

Exercise programme modification will be based on a subjective and/or objective re-assessment and include either:

- i) progression of prescribed exercises if tolerated with minimal pain and discomfort,
- ii) maintenance if tolerated but some moderate pain and discomfort, or
- iii) reduction in frequency, duration, intensity or modification of exercise type if not tolerated or adhered to.

The exercise programme will be supervised during each treatment appointment and may be progressed at subsequent treatment appointments according to observed changes in presentation, modelled on successful exercise interventions in our earlier trials.^{48,49} A record of the exercise prescription will be recorded on the Intervention Details CRF for each treatment appointment. Adherence will be encouraged by provision of written individualised exercise sheets (online or paper) and use of a paper-based diary to discuss with the clinician during subsequent treatment appointments. The diary will be collected at 12 weeks.



Figure 7: Summary flow chart of exercise prescription

4.1.3 SMA booklet plus prefabricated foot orthoses (SMA-orthoses)

Participants randomised to SMA-orthoses will be given the SMA booklet at the initial treatment appointment. The treating clinician (a physiotherapist or podiatrist) will assess foot posture using the Foot Posture Index-6 (FPI-6)⁵⁰ and select the appropriate orthotic device according to the degree of static rearfoot eversion and body weight (see Figure 8 for the foot orthoses and body weight

algorithm). These data will be recorded on the CRF by the treating clinician. If deemed clinically relevant to PHP, this can include assessment of hip position (anteversion/retroversion) and knee position (genu varus/valgus).

Protocol for prescription of prefabricated foot orthoses

The assessment protocol for the prescription of foot orthoses includes two key components to guide the physiotherapist/podiatrist towards the most appropriate foot orthosis prescription.

- 1. A main driver in influencing the rearfoot posting component of the foot orthosis prescription will be the rearfoot posture component of the FPI-6⁵⁰. This will be assessed by the podiatrist/physiotherapist when the participant is in a relaxed stance position. The podiatrist/physiotherapist will observe whether the calcaneus is inverted, vertical, everted, or highly everted (see Figure 8).
- 2. Assessment of body weight will also influence the selection of the appropriate orthotic device (greater body weight resulting in selection of a device with a higher material density).
- 3. Each foot orthotic device will be fitted according to the size of the participant's foot using the orthotic device shells of various sizes.
- 4. Clinicians will select the appropriate first choice device with appropriate rearfoot medial posting dose (see Figure 8) in place and assess the participant for correct size of orthotic device (weight-bearing and non-weight-bearing fit-to-foot).
- 5. Clinicians will then check the fit of the orthotic device to the shoe (fit-to-shoe).
- 6. Tolerance will be evaluated by asking the participant if they are happy with the comfort and fit of their orthotic device (tolerance). If the participant is not happy with comfort or fit, the clinician may choose to taper the shell density and/or the dose of rearfoot posting.
- 7. Foot orthoses should be prescribed for both feet according to the intervention protocol which is driven by participants' foot posture and bodyweight, regardless of whether the heel pain presentation is unilateral or bilateral.

Foot orthoses choice

In the development of our pilot and feasibility trial³³, our patient and clinician advisory group identified desirable characteristics of an orthotic intervention protocol including:

- (1) an element of patient choice between different devices based on comfort and fit
- (2) scope for device adjustment/tailoring by the clinician to provide the desirable level of rearfoot posture/functional control for individual symptoms.

As a result, we developed a pragmatic foot orthosis intervention algorithm, which includes the following devices:

- Vectorthotic® (firm density shell),
- Salfordinsole[™] Firm (medium to firm density shell),
- Salfordinsole[™] Flex (low-medium density shell).

These devices are prefabricated and modifiable with the use of 'click-in' or adhesive additions (medial rearfoot posts) which can be used to change the level of pronatory control, as well as patient comfort and therefore potentially influence adherence. A range of shell material densities allows for a more supportive orthotic for participants with a higher bodyweight.⁵¹ At present, there is little evidence to suggest that one brand/type of foot orthoses is more effective than another for the management of PHP.^{22,52} Participants will be instructed how to fit the device and advised to wear it for one hour per

day, gradually increasing by one hour per day up to at least four hours per day, and given an individualised and detailed orthosis information sheet. Participants will be followed up for up to 6 treatment appointments over 12 weeks at the discretion of the treating clinician (face-to-face appointments prioritised if possible, or virtual, telephone, if required). The orthosis can be altered during subsequent consultations according to participants' self-reported tolerance or clinical presentation. Adherence will be encouraged by a diary to capture use and facilitate discussion with the clinician. The diary will be collected at week 12. The foot orthosis may be changed or altered during subsequent clinical treatment appointments according to participants' self-report of tolerance or clinical presentation.



BW: Body weight.

Figure 8: Summary flow chart of orthoses prescription

Foot orthoses details

Materials: Vectorthotic devices[®] - semi-rigid polypropylene shell with a closed cell polyethylene cover and polypropylene 2°, 4° and 6° rearfoot posts, Salfordinsole[™] shells and 4° rearfoot posts, both Sure Step-Control[™] thermoplastic elastomer material.

Supplier details: Participants randomised to receive foot orthoses should receive a trial orthotic device. These will be supplied free of charge by Keele CTU in accordance with the Supply Instructions provided in the Investigator Site File (ISF).

Storage and consignment stock: Orthotic devices should be stored in accordance with manufacturer's recommendations and as outlined in the supply instructions provided in the trial information. A consignment stock of orthoses will be sourced and purchased by a member of the research team and delivered to each site. This will ensure that participants can be provided with the appropriate orthoses, as required. The Principal Investigator at each site is responsible for consignment stock monitoring and replenishment. At the end of the trial, any leftover stock will be returned to Keele CTU.

<u>4.1.4 SMA booklet plus individualised exercises and prefabricated foot orthoses (SMAcombined)</u>

In addition to the SMA booklet provided at the initial treatment appointment, participants randomised to SMA-combined will receive both individualised exercises and prefabricated foot orthoses interventions as described above and delivered over up to 6 treatment appointments, over 12 weeks, by a physiotherapist or podiatrist. Duration of appointments will be the same as in the SMA-exercises and SMA-orthoses arms.

4.2 Interventions/Treatments general treatment appointment procedures

Participating physiotherapy and podiatry services responsible for delivering the interventions will be expected to maintain an ISF of essential trial documentation, which will be provided by Keele CTU, and keep copies of all completed CRFs for the trial in accordance with the Sponsor-Site Agreement.

Clinical trial data will be recorded by clinicians who are taking part in the trial. All clinicians will be trained in accordance with the protocol on how to record trial-specific data. Data will be collected on either paper CRFs or electronic CRFs depending on site capability. Paper CRFs will be collected by a trial team member, posted, faxed or emailed to Keele CTU as agreed with the site. Electronic CRFs will be entered directly onto the REDCap trial database by the clinician. Only the participant's trial number plus date of birth, initials and sex at birth will be included on CRFs. Where paper CRFs are sent to Keele CTU, the trial participating services are responsible for redacting all other personal identifiable data prior to sending. Following receipt of paper CRFs that are sent, Keele CTU will contact trial sites to resolve any missing or discrepant data queries relating to clinical data in accordance with Keele CTU procedures.

For all three clinician-supported intervention arms (SMA-exercises; SMA-orthoses; SMA-combined), participants who do not attend (DNA), or who are unable to attend (UTA) their physiotherapist/podiatrist appointment should be reappointed for the duration of the trial and CRFs should be completed to note the DNA/cancellation. This will take place flexibly in conjunction with participating services. Clinicians will be trained to deliver both exercise and orthosis interventions and will be notified which arm the participant has been randomised to in advance of their first treatment appointment. Clinicians will be required to complete a treatment CRF for each patient appointment. Table 1 summarises each intervention arm.

Self-management advice	Self-management advice <i>plus</i> individualised exercises	Self-management advice <i>plus</i> prefabricated foot orthoses	Self-management <i>plus</i> individualised exercises and prefabricated foot orthoses
Advice and information booklet	Advice and information booklet Up to 6 sessions over 12-week treatment period Individually tailored and written exercise programme Focus on foot/ankle specific stretching and strengthening +/- lower limb muscle groups Individualised exercise programme that is progressed Clinical supervision of exercise programme Exercise diary	Advice and information booklet Up to 6 sessions over 12-week treatment period Individually tailored pre-fabricated foot orthotic device Device adjustment/ tailoring as required Clinical supervision of orthotic device use/ tolerance Orthosis diary	orthosesAdvice and information bookletUp to 6 sessions over 12-week treatment periodIndividually tailored and written exercise programmeFocus on foot/ankle specific stretching and strengthening +/- lower limb muscle groupsIndividualised exercise programmeClinical supervision of exercise programmeIndividually tailored pre-fabricated foot orthotic deviceDevice adjustment/ tailoring as required
			orthotic device use/ tolerance Exercise & orthosis diary

Table 1: Summary table of interventions

4.3 Trial Training

Treatment packages for those participants randomised to interventions which include exercises or foot orthoses treatments, will be delivered by participating physiotherapy and podiatry sites. Participating services will be those who have agreed to participate, have local management approval, have been trained in trial specific interventions and procedures, and are accessible to patients registered at the

GP practices participating in the trial. A research clinician (e.g. physiotherapist, podiatrist, nurse) will be trained to perform telephone screening of potential participants, confirm eligibility and obtain consent. Trial administrators will also be trained to support site processes (e.g. booking participants into treatment appointments) as required.

Physiotherapists and podiatrists: Treating physiotherapists and podiatrists (at least NHS Agenda for Change Band 5) will participate in training workshops (2- 3 days) with the research team prior to the start of recruitment and treatment (hybrid remote and face-to-face delivery or remote delivery only if the covid-19 pandemic requires). The workshop(s) will be based on the training provided in our pilot and feasibility trial. The focus of the training will be on carrying out standardised assessment, delivery of the exercises and orthosis interventions in line with the agreed protocols (including remote delivery if the covid-19 pandemic requires) and supporting participants to adhere to the interventions. The training also will cover completion of all trial paperwork requirements including case report forms to assess intervention fidelity, Good Clinical Practice as applicable to research, maintenance of the site file and trial records, and reporting of serious adverse events, serious breaches, adverse events and protocol non-compliance.

In the pilot and feasibility trial, participants in the three clinician-supported intervention arms attended a mean of 2.3 consultations whereas the protocol permitted up to 6 sessions and adherence to exercises in the SMA-exercises arm was lower than adherence to orthoses in the SMA-orthoses arm (and adherence to exercises and orthoses in the SMA-combined arm). In the clinician training programme, we will therefore emphasise the importance of supervision and progression of exercises by utilising the available number of treatment sessions and strategies to improve adherence to exercises such as encouraging completion and review of exercise diaries. Throughout the training programme, we will emphasise the key characteristics of the therapeutic alliance that facilitate exercise adherence, including agreement on goals and tasks (goal setting), clear communication, a sense of connectedness, positive feedback, genuine interest, individualised care plans (including exploring specific barriers to exercise adherence), trust in the clinician and feeling empowered.^{53,54}

The training will be supplemented by a comprehensive manual, providing clear treatment protocols and guidance for completing trial paperwork. Case Report Forms completed by clinicians to capture data on intervention delivery will be regularly audited against the treatment protocol to examine if the interventions are being delivered in accordance with the protocol. Individual feedback will be given to clinicians if required. Refresher training will be offered where needed. Additional training on remote delivery of interventions will also be provided if the covid-19 pandemic requires.

5 TRIAL SETTING

Participants will be recruited from NHS general practice, physiotherapy and podiatry services and from self-referral from within the community following an awareness raising campaign, within England and Scotland. Interventions will be delivered in TREADON treatment appointments that will be integrated into routine NHS physiotherapy and podiatry services. This will ensure sampling from a heterogeneous population spanning a range of area-level deprivation/affluence and a mix of urban, semi-rural and rural areas.

6 ELIGIBILITY CRITERIA

The target population is adults aged 18 years and over with PHP. Our inclusion/exclusion criteria are broad, and thus representative of a primary care population (Table 2). Only participants who have symptoms unlikely to be attributable to PHP, current/recent experience of the interventions, or for whom the interventions would be contraindicated/inappropriate will be excluded. There are no exclusions based on language and we will utilise local interpreter services (e.g. NHS, local authority) at all stages of data collection to ensure the trial is accessible to non-English speaking populations.

Eligibility for the trial has been informed by our pilot and feasibility trial.³³

Table 2. Eligibility criteria (assessed by telephone eligibility screening)

Inclusion criteria	
Adults aged 18 years and over	

Self-reported localised pain under the heel which is worst when standing after rest and after prolonged weight-bearing

Symptom duration at least 4 weeks with pain ≥2 (0-10 Numeric Rating Scale [NRS])

Access to a mobile phone that can send/receive SMS text-messages or a landline telephone

Able and willing to participate and provide informed consent

Exclusion criteria

Inflammatory arthritis, systemic lupus erythematosus, gout, fibromyalgia

Heel pain of neurologic cause e.g. tarsal tunnel, entrapment neuropathy, radiculopathy

Serious pathologies requiring urgent medical attention (e.g. trauma, tumour, infection)

Current treatment or treatment in the last 3 months by a physiotherapist or podiatrist for PHP, or currently using a contoured foot orthosis

Previous surgery or awaiting surgery on the affected foot

Corticosteroid injection into the affected foot in the last 3 months

Extracorporeal shockwave therapy to the affected foot in the last 3 months

Unlikely to tolerate the interventions (e.g. allergies to common orthotic device materials) or unable to attend for treatment

7 TRIAL PROCEDURES

7.1 Recruitment

Patient identification

Potential participants will be identified based on the three methods that were employed successfully in our pilot and feasibility trial: 1) general practice consultation, 2) retrospective review of general practice medical records, 3) population survey.³³ In addition, and to further optimise recruitment, we will 4) screen referrals to NHS physiotherapy and podiatry services and, on the advice of our PPIE group, 5) identify participants through community advertising and self-referral. All recruitment arms will proceed in parallel until the required sample size is reached. See Figures 1-6 above for summary flowcharts of the five methods.

For recruitment methods 1-4, participating sites will either upload to REDCap or send a file, via secure transfer, to the Keele CTU team for upload upon sending survey invites. This will be on a regular basis (approximately weekly) for methods 1 & 4. Methods 2 & 3 will be sent when the search and mail out has been completed. These files are to include the Organisational Code, invite date and method of invitation for each patient sent an invitation to take part in the trial, plus NHS number, anonymised identifier, age, and sex where IT systems allow. When screening questionnaires are received into the CTU (online or paper) the information from their consent to take part in the trial will be matched to the NHS number from the import to ensure that the correct patient is taking part in the trial and allows reminders to be sent from the sites to the correct patients (non-responders only). NHS numbers of those that do not respond to the trial invitation will be deleted.

<u>Method 1 – Rolling monthly identification - General practice consultation: Foot/ankle pain</u> <u>consulters</u>

Clinical System Protocol (Preferred method)

Potential participants will be identified through routine primary care consultation (face-to-face or telephone) about PHP at their general practice. Where possible, a study designed general practice clinical system protocol will automatically identify, screen and code potential participants aged 18 years and over. The protocol will trigger when specific SNOMED or Read codes are entered into a consultation, these codes are based on foot/ankle pain related diagnostic or symptomatic Read or SNOMED codes that were successfully used to identify adults with PHP in our pilot and feasibility trial.³³ As in our pilot and feasibility trial, a broad range of symptom codes will be used because foot/ankle consultations are often not coded with specific diagnostic labels such as PHP.⁵⁵ The protocol will automatically exclude those patients with potentially serious pathology (e.g. inflammatory arthritis, as listed in the exclusion criteria), potentially vulnerable patients. The clinician triggering the protocol will be asked to confirm if the patient is suffering from plantar heel pain and therefore eligible to receive further information. A clinical system search will be manually run approximately every 4 weeks to identify these eligible patients.

Clinical System Search

In practices where using a Clinical System protocol is not possible, a study designed clinical system search will be run to identify patients who have had a qualifying foot/ankle consultation in the last 4 weeks. For consistency, the search will utilise the exact same SNOMED/Read code lists that the protocol uses to identify (trigger) and for the automated exclusions. A practice clinician will then perform a manual screen of this list to confirm if plantar heel pain present.

Patient contact

Approximately every 4 weeks, the GP practice will send potentially eligible patients identified in the last 4 weeks, identified as above, an SMS text-message containing a link URL to the online screening survey pack(including the survey participant information leaflet, online consent to contact reply form and the screening survey), with a reminder SMS text-message to be sent within 7 days after the initial message.

Patients who do not have an active mobile number, or who have opted out of SMS contact from their GP practice, will be sent the screening survey pack by post (an invitation letter from their GP, survey participant information leaflet, postal consent to contact reply form and screening survey with a prepaid return envelope) via Docmail, which is a standards-compliant hybrid mail service, providing document management and ISO 27001 secure mailings.

The screening survey will ask respondents to indicate the location of pain on a validated foot manikin^{56,57} (© The University of Manchester 2000. All rights reserved) and complete questions regarding foot pain, demographic details and eligibility. For those completing the foot manikin online, the participant will select the location of pain on the foot manikin. For those completing a paper survey, the participant will shade directly the location of pain on the manikin, which will be read by overlaying a reliable coding template previously used in our foot pain research to identify PHP.⁵⁷

Non-responders will be sent a reminder approximately four weeks after the initial invite. Pilot and feasibility trial recruitment rate: 6/10,000 practice population.³³

Method 2 - Retrospective (1 year) general practice search: Foot/ankle pain consulters

The GP practice will run a study designed clinical system search to identify adults aged 18 years and over and have a SNOMED/Read coded consultation for foot/ankle pain in the preceding year. The clinical criteria of the search for inclusion and exclusion will utilise the same code lists used in Method 1.

The GP practice will then contact eligible screened patients via the SMS and Docmail methods outlined in method 1 above.

Non-responders will be sent a reminder approximately four weeks after the initial invite. Potential participants who appear eligible and who consent to contact will be sent a full trial information pack either online or postally. Pilot and feasibility trial recruitment rate: 2.6/10,000 practice population.³³

Method 3 – General practice population survey

All adults registered at selected general practices will be contacted to identify those who had heel pain but had not consulted in the preceding year. Our data indicate that only 43% of people with PHP consult their GP². The GP practice will run a clinical system study designed search to identify all registered adults (\geq 18 years of age and exclude potentially vulnerable patients), and GPs / primary care clinicians will screen lists to exclude potentially vulnerable patients. Patients will be contacted, and non-responders followed up, as described in Method 1 above. Pilot and feasibility trial recruitment rate: 27/10,000 practice population.³³

Method 4 - Physiotherapy and podiatry referrals

To capture potential participants who have either been referred from primary care or self-referred to physiotherapy/podiatry, the NHS trust/Health Board will screen referrals against eligibility criteria to identify adults with PHP. Where IT systems allow, identified patients will be sent an SMS text-message from the physiotherapy/podiatry service providing a link URL to the online screening survey pack (including the survey participant information leaflet, online consent to contact reply form and the screening survey). Patients who are unable or unwilling to complete this online will be able to opt for postal completion. Where IT systems do not have SMS function, patients who do not have an active mobile number will be sent the screening survey pack by post from the physiotherapy/podiatry service or via Docmail. Non-responders will be sent a reminder approximately four weeks after the initial invite.

Method 5 - Self-referral from the community

Our PPIE group told us that not all individuals with PHP will consult about it. Therefore, to capture nonconsulters, adults with PHP will be able to self-refer for consideration for the trial. Local communities around each participating NHS service will be provided with information about the trial via an awareness raising campaign. Informed by our previously successful awareness raising campaign in the HTA-funded EASE Back feasibility and pilot trial⁵⁸ and PROP OA trial⁵⁹ and PPIE feedback, information directing individuals to the research team's online/telephone contact details will be shared via social media (Facebook and Twitter), a trial website, local radio and newspapers, and flyers, posters in general practices, physiotherapy and podiatry practices, and community settings. Where possible, automated check-in screens for patients attending general practices, and waiting room display screens, will display information about the trial. Potential participants who register their interest via the trial website will be directed to the online screening survey pack (including the survey participant information leaflet, online consent to contact reply form and the screening survey).

Those contacting Keele CTU directly will be given the option to complete the screening survey either online or via post. Those wishing to complete online will be provided with a link URL to the online screening survey pack. Patients who are unable or unwilling to complete this online will be able to opt for postal completion. Non-responders will be sent a reminder approximately four weeks after the initial survey invite.

Duplicate invitations

All communications to the patients identified via methods 1-4 will be from the GP practice or NHS trust/Health Board. Whereby a patient is identified by their general practice via methods 1, 2 or 3, systematic searches will be conducted to ensure that potentially eligible participants will not receive duplicate invitations if identified via any of these methods. Invitation correspondence with potential participants will include a statement that we cannot completely eliminate the possibility of duplicate invitation.

Full trial information pack

Participants will be asked to indicate their preferred method of contact (for trial information and questionnaires), either as post or electronically, on their 'consent-to-contact' form. To facilitate online methods, email address and mobile number will be requested from those who prefer electronic contact.

The full trial information pack (which will be pre-coded with a participant trial ID number) will be sent to participants through a link URL or by post, depending on their preference, from Keele CTU and will include:

- a cover letter and/or invitation to consider participation in the trial
- participant information leaflet (PIL)
- trial consent form
- baseline questionnaire (pre-treatment questionnaire)
- pre-paid return envelope (as required).

This will allow potentially eligible participants to consider the trial in detail in their own time.

Telephone screening

Potential participants sent a full trial information pack by any of the five recruitment methods will be called by a clinical researcher, within 2-5 working days after sending the full trial information pack, to explain the trial, confirm eligibility and facilitate consent to participate. Up to five attempts will be made to contact the patient, including at different times of the day. Where no contact is made this will be recorded.

The completed trial consent form and baseline questionnaire will be returned to Keele CTU online or by post.

During this telephone contact the clinical researcher will check and confirm all eligibility criteria in order to ensure that only those who meet these criteria are recruited to the trial. A Telephone Eligibility Screening Form for all patients considered for inclusion will be completed. Information regarding age and sex at birth will be retained as unidentifiable variables from the Eligibility Screening Forms. Documented reasons for ineligibility or declining participation will be monitored by the Keele CTU as part of a regular review of recruitment progress. Screened patients who are not randomised either because they are ineligible or because they decline participation will also have the following information recorded:

- the reason why the patient is not eligible for trial participation, OR
- the reason for declining if eligible. Patients are not required to give a reason, but if a reason is ascertained it will be recorded.

If the individual has agreed to further contact by the research team but has not provided a contact telephone number, an adapted invitation letter will be sent by postal mailing, stating that the research team do not have their contact number and if they are interested in participating to either contact the trial team using the information provided or return their telephone contact details on the form and free post envelope provided. If no contact is made within 10 working days from the date the letter was sent, then the trial team will not contact them again.

Prior to receiving verbal and subsequently written informed consent to trial participation, the clinical researcher must have established the patient's eligibility for inclusion (see section 6). The trial will be explained, as per the PIL, over the telephone for the patient to consider. This will include information about the rationale, design, treatment packages and follow-up requirements of the trial. The clinical researcher will address any concerns or questions. Prior to receiving this telephone call, the patient will have received a copy of the PIL for them to read in their own time and consider.

Those patients who have verbally consented to take part in the trial will be asked to complete the baseline questionnaire included in the trial pack and return it to the research team along with a signed and dated consent form in the pre-paid envelope provided (or equivalent online correspondence). On receipt at Keele CTU, the authorised trial administrator will check it for completeness before the participant proceeds to the randomisation stage. Online correspondence checking will be performed by automated processes. Participants who are eligible and have returned their consent form will be contacted by phone if they do not return their baseline questionnaire after seven working days of the original consent phone call date, to remind them to return their required correspondence. As described above in this section, where phone contact is not made or if after a further 7 days from the reminder phone call, the patient still has not returned their required correspondence, a letter, through a link URL or by postal mailing, will be sent to remind them to return their questionnaire.

7.2 Consent

Consent to full trial participation

A research team member will gain verbal informed consent from willing eligible patients by reading through each section of the consent form explicitly and clarifying each point the individual needs to confirm.

Those who wish to take part in the trial will be asked to sign and date the consent form included in their baseline pack and return to the research team either in electronic-consent online format or in the pre-paid envelope provided to those who opt for paper-based involvement. Patients who return an incomplete paper consent form will be sent a letter asking them to complete the consent form, a copy of their consent form with the incomplete section(s) highlighted and a pre-paid return envelope. Recruitment will be complete when the signed consent form is received at Keele CTU. This recruitment method is identical to a recent successful trial,⁶⁰ and the TREADON pilot and feasibility trial³³, with additional modification to incorporate an online correspondence option.

Patients who decline to participate during the telephone call will be thanked for their time and advised that they should contact their GP or referring physiotherapy/podiatry service regarding further care if required. The reason for declining (if given) will be recorded for anonymous reporting purposes and personal details will be cleared from the contact database in accordance with Keele CTU procedures.

The clinical researcher taking part in the informed consent process will have received appropriate training and will be authorised in the trial delegation log and permitted to take informed consent. The right of the participant to refuse consent without giving reasons will be respected. Furthermore, the participant will remain free to withdraw from the trial at any time without giving reasons and without prejudicing any further treatment.

The signed (online authorised by patient) and dated trial consent form will be retained by Keele CTU along with a record of the consent process detailing the date of verbal consent and will be separated from any research clinical data.

Follow-up of people for whom the consent form is not signed and returned

Individuals who are eligible and have verbally provided consent by phone will be contacted by phone if they do not return their consent form and/or baseline questionnaire after seven working days of the original consent phone call date, to remind them to return their online or offline paperwork. Where phone contact is not made or if after a further seven days from the reminder phone call, the patient still has not returned their paperwork, a letter will be sent, through a link URL or by postal mailing, to remind them to return their online/offline paperwork.

Informed Consent Responsibilities

The Chief Investigator (CI) retains overall responsibility for the informed consent of participants through Keele CTU. The CI will also ensure that any person delegated responsibility to participate in the informed consent process is duly authorised, trained and competent to participate according to the ethically approved protocol, principles of GCP and Declaration of Helsinki 1996. Such delegated

responsibilities will be recorded in a trial delegation log. Informed consent will be obtained prior to the participant undergoing procedures that are specifically for the purposes of the trial and are out-with standard routine care at the participating site. The right of a participant to refuse participation without giving reasons will be respected.

Where a participant is required to re-consent or new information is required to be provided to a participant, it is the responsibility of the CI to ensure this is done in a timely manner and according to any timelines requested by Keele CTU.

Loss of Capacity Following Informed Consent

Where valid, informed consent is provided by the participant and the participant subsequently becomes unable to give on-going informed consent by virtue of physical or mental incapacity, the initial consent provided remains legally valid and endures. Participants who lose capacity after informed consent has been obtained and are unable to continue with protocol treatment or questionnaires will be withdrawn from further active trial participation.

7.3 Randomisation

Following confirmation of eligibility, receipt of a correctly completed and signed consent form and completed baseline questionnaire, participants will be randomly allocated to an intervention. Randomisation will be conducted by an authorised administrator (at site or CTU, depending on service preference) via REDCap. This is a secure web-based data collection system that uses a randomisation module; the randomisation sequence will be computer-generated (random block randomisation, stratified by identification method and geographical region). Emergency telephone backup will also be available. Allocation will be concealed from the blinded researchers, as outlined in section 7.4. Participants will be randomised on an equal 1:1:1:1 basis to one of the following interventions:

- (i) Self-management advice (SMA) booklet (control arm)
- (ii) SMA booklet plus individualised exercises (SMA-exercises)
- (iii) SMA booklet plus pre-fabricated foot orthoses (SMA-orthoses)
- (iv) SMA booklet plus individualised exercises and pre-fabricated foot orthoses (SMA-combined).

The following information will be required for randomisation:

- Participant details, including initials, sex at birth and date of birth
- Participant trial ID
- Name of person undertaking randomisation
- Confirmation of eligibility
- Confirmation of written informed consent and date
- Completed baseline questionnaire including participant recorded pain score
- Identification method and geographical region (stratification variables).

Authorised access codes and passwords will be required to access REDCap. On successful access to the system the trial administrator will be notified of the participant's treatment allocation.

All participants who are randomised will be informed of their allocation through a link URL or in writing by postal mailing and receive a high quality SMA booklet. Participants in the SMA control arm will be posted an SMA booklet. Participants in the three treatment arms will be given an SMA booklet at their first treatment appointment. Where participants have been allocated to a treatment package which includes exercise and/ or orthotic device intervention, a coordinator member of the trial team will then liaise with the appropriate service and a first appointment for each trial participant will be made. This will take place through normal routine service channels. It is anticipated that the service will be responsible for notifying the participant of the date, time and location of their first appointment. We aim

to agree a maximum wait for treatment start of no more than 2 weeks following randomisation with participating sites. It is also anticipated that subsequent appointments will be managed through the normal routine service channels. Patients who DNA/UTA appointments will be offered further appointments as needed as part of the trial. The GP of each trial participant will be sent a letter to confirm that their patient is taking part in the research trial. Participants allocated to the SMA booklet control arm will not be required to attend for any trial-specific clinical intervention.

Selection bias at recruitment will be avoided by the use of the following methods; all patients will receive identical trial information and questionnaires minimising the threat to participation bias from the different identification routes. A clinical researcher will contact patients to discuss eligibility regardless of identification route and will not take part in treatment allocation. Allocation concealment will be achieved by separating the processes and individuals involved in determining treatment allocation and treatment delivery and through the use of random permuted blocks.

7.4 Blinding

It will not be possible to blind participants and treating clinicians (physiotherapists and podiatrists) to treatment allocation. Research staff involved in data collection and analyses (including the trial administrator overseeing collection of SMS text-messages, follow-up questionnaires and minimal data collection; data entry administrator; and a blinded lead statistician) will remain blind to treatment allocation until all data collection up to 12-month follow-up has been completed and the main intention-to-treat analyses of clinical effectiveness at 12 months performed. Statistical analysis of the primary outcome including interim analysis will be carried out by an (unblinded) trial statistician and separately by the (blinded) lead statistician with consensus agreement, and independently verified by a (blind) external statistician collaborator (with expertise in adaptive trials) to ensure the integrity of the trial evaluation. To ensure these research staff remain blind to treatment allocation, treatment arms in relevant databases will always be stored as a unique code and code break will be held within the REDCap database and a paper copy stored securely (only accessible to the Software Development Team; hidden from others who have access to the REDCap database).

7.5 Baseline data

The following forms will be completed prior to treatment:

- Informed consent form
- Eligibility Screening CRF
- Informed consent form for randomisation and intervention
- Participant baseline questionnaire.

7.6 Trial assessments

Table 3. Schedule of enrolment, interventions and assessments

		Enro	Enrolment Randomisation				Post-randomisation				
Measure	Description	Screening survey	Enrolment	Baseline pre-treatment	Random allocation and initial treatment	weekly text/call follow-up for 12 weeks/then monthly	Up to 6 clinician follow-ups	Weekly Diary	12 week follow-up	6 month follow-up	12 month follow-up
Enrolment											
	Telephone screening		\checkmark								
	Informed consent	\checkmark	\checkmark	\checkmark							
	Random allocation				\checkmark						
	Initial treatment visit				\checkmark						
Participant descriptors											
Demographics	Date of birth, sex at birth	✓		✓					✓	✓	✓
Ethnicity	Ethnic origin			\checkmark							
Body mass index	Height and weight			\checkmark							
Typical activity, last week	Time on feet, hours per day			\checkmark							
Previous PHP episodes	Number of episodes			\checkmark							
Previous treatments received	Type and number, including exercise/orthotic device treatments: 2 questions			✓							
Current PHP episode description	Duration and severity: 2 questions	\checkmark		\checkmark							
	Laterality	\checkmark		\checkmark					\checkmark	\checkmark	\checkmark

Preference for treatment intervention	Patient self-report: 2 questions	✓							
Clinical assessment									
Foot Posture Index-6	Foot posture		\checkmark						
Interventions									
Self-management advice (SMA)			√						
Self-management advice booklet plus individualised exercises (SMA-exercise)			\checkmark		√				
Self-management advice booklet plus prefabricated orthoses (SMA-orthoses)			√		√				
Self-management advice booklet plus individualised exercises and prefabricated orthoses (SMA-combined)			✓		✓				
Intervention adherence									
						✓	✓	✓	✓
Intervention credibility and s	satisfaction								
							✓	✓	√
Adverse events related to in	terventions								
						√	✓	✓	√
Clinical outcome									
Plantar Heel Pain	Average plantar heel pain over the last 7 days using numeric rating scale (0- 10 NRS) '0' denoting no pain and 10 worst pain imaginable	✓		√			✓	✓	✓
	Presence of pain in the heel; yes or no	\checkmark					\checkmark	\checkmark	\checkmark
	First step pain (NRS 0-10)	\checkmark					\checkmark	\checkmark	\checkmark
IRAS 3	14272_TREADON main trial protocol_v2.0 08Dec2022	2		Page 37	of 61				

	Global Impression of change score, 6 point scale			✓	✓	✓
	Pain self-efficacy questionnaire, 10 items, 6 point scale		\checkmark	√	\checkmark	✓
	Brief illness perception questionnaire, 8 items, 10 point scale		\checkmark	✓	\checkmark	~
Foot pain	Foot Function Index pain subscale, 9 items, 10 point Likert scale		\checkmark	✓	\checkmark	~
Presence and location of pain in the foot	Foot Manikin (© The University of Manchester 2000. All rights reserved)	\checkmark		√	\checkmark	✓
Foot Function	Foot Function Index disability (9 items) and activity limitation subscale (5 items),10 point Likert scale		~	√	✓	~
Health related quality of life	EuroQuol:EQ5D-5L		\checkmark	\checkmark	\checkmark	\checkmark
Healthcare costs						
Employment	Current employment status		✓	✓	✓	✓
Performance at work	How performance at work is affected (NRS 0-10)		\checkmark	✓	✓	✓
Work loss (absenteeism/presenteeism)	Number of days lost		✓	✓	\checkmark	✓
PHP Healthcare utilisation	Use of prescribed or over-the-counter medications or interventions e.g. foot orthoses, heel pads			✓	~	✓
	Hospital investigations, treatments and use of private healthcare			√	\checkmark	~

7.7 Follow-up assessments

Outcome collection methods and time points

Text-messages weekly for 12 weeks, then monthly to month 12

All participants will be asked to respond to mobile phone SMS text-messages which will be automatically sent each week for 12 weeks, and monthly thereafter to month 12, to collect pain severity scores. Patients who do not wish to use text-messaging will be offered the option of brief telephone calls at the consent stage.

The research team member will explain this process over the telephone, when gaining informed consent at the start of the participation. Non-responders to the initial text-message will be sent a further automated text-message reminder 24 hours later, those receiving phone calls will receive another call the next working day. Non-response to this text/phone call will be recorded and no further contact at this attempt made.

96% of UK adults have a mobile phone (https://cybercrew.uk/blog/smartphone-usage-statistics-uk/) and previous research shows that weekly text-messages (followed by a telephone call for the first text-message that is not answered) had an 83% mean response rate.^{49,61}

Weekly adherence Patient Diary for 12 weeks

All patients randomised to one of the three clinician-supported intervention arms will be asked to keep a weekly paper-based diary over the 12 weeks of the intervention. This will capture aspects of adherence, general patient engagement with the intervention and any self-reported adverse events. The diary can also be used by patients to aid discussions with their treating clinicians. Patients will be asked to return the diary to Keele CTU at 12 weeks, using a pre-paid envelope provided. If after 13 weeks, the patient still has not returned their diary a reminder will be sent by SMS text-message or by postal mailing.

Questionnaire at 12 Weeks, 6 months and 12 months

All participants will be asked to complete a self-report questionnaire 12 weeks, 6 months and 12 months after randomisation. The questionnaire (online or paper) will be sent to participants by Keele CTU at 12 weeks, 6 months and 12 months' post randomisation. Where a questionnaire has not been returned to Keele CTU within 10 days from link URL in SMS text-message/mail despatch, a reminder text-message, or postcard will be sent. Where a questionnaire is still not returned following this, a second questionnaire will be sent 5 working days after the text-message (or postcard). For non-responders, minimum data collection (MDC) to capture the primary outcome measures will be undertaken by a Keele trial team member by telephone approximately 3 weeks after the initial SMS text-message invite/mailing, during this phone call if participants decline to provide MDC over the phone they can be offered the choice to receive an online or postal MDC questionnaire. Where a participant is not able to be contacted by telephone, a MDC questionnaire will be sent through a link URL in an SMS text-message or by post. Participants will be asked to return the completed questionnaire (together with the patient diary at 12 weeks) to the Keele CTU, either online or using the pre-paid envelope.

7.8 Withdrawal criteria

Participants may withdraw from the trial at any time without giving reasons and without prejudicing any further treatment. Any information provided up to the point the participant withdraws will be included in the dataset and used unless the participant asks for their data to be destroyed.

Any discontinuation of exercise or orthoses interventions for any reason is not a reason for withdrawal from the trial.

7.9 Storage and analysis of samples

All data will be stored and anonymised in line with Keele CTU Quality Management Systems.

7.10 End of trial

The trial end is at the point at which the trial database is locked. CRF data will have been received by the data management team at Keele CTU and any data queries will have been resolved. Copies of CRFs will remain at each participating site. The Chief Investigator will notify the REC of the end of the trial within 90 days of trial completion.

8 STATISTICS AND DATA ANALYSIS

8.1 Sample size calculation

Up to 696 participants will be required to detect a small-to-moderate standardised between-group effectsize of 0.3 (equating to a minimum clinically important difference of 0.8 in average pain NRS over 6-12 weeks follow-up with anticipated standard deviation 2.7) in at least one clinician-supported intervention arm against the control arm (SMA booklet) with 90% probability, while controlling the probability of the overall type I error at 5% (one-sided).^{62,63} The maximum sample size is based on equal allocation of participants, on the assumed, repeated measures correlation of 0.7. baseline-outcome correlation of 0.5⁶⁴, 20% loss to follow-up, and 70% response to SMS text-messages (informed by our pilot trial and previous trials). This accounts for multiple comparisons of each of the intervention arms to the control and one formal interim analysis upon the primary outcome being observed for 348 participants (87 per arm).⁶⁵ The total sample size of 696 also represents the maximum possible sample size under the 2stage MAMS adaptive trial design; the minimum sample size being half of this if the trial is stopped (all three clinician-supported intervention-control interim evaluations yield between-group differences below the futility bound or at least one beyond the superiority bound) after the interim analysis; hence, the actual sample size will be in the range of 348 to 696. Extensive (>100,000) simulations indicate that the expected sample size varies between 415 and 532 participants. For logistical reasons, recruitment will not be suspended whilst the formal interim analysis is undertaken; to maintain momentum with recruitment, the trial will carry on recruiting and hence we anticipate, due to the expected lag of around 4 months (12week data follow-up plus 4 weeks for data transfer, checking and interim analysis), approximately 100 further participants will have been recruited by the time of the interim decision-making. Hence, taking this lag into account, the revised minimum sample size would be about 448 and the expected sample size range would be about 490-610. If the efficacy boundary is crossed at the time of the interim analysis and the trial is stopped earlier, the conclusion on the efficacy of this treatment will be based on the test statistic (and the corresponding efficacy boundary value given in Section 8.3.2 below) at the time of the interim analysis and will not be affected by the data from the participants recruited after the interim analysis. The data from the stage-2 participants, however, in this case, will be included in the primary and secondary analysis to obtain more accurate treatment effect estimates.

8.2 Planned recruitment rate

Estimated recruitment rates: Based on our pilot and feasibility trial³³, we anticipate needing 45,000 adults to complete the screening survey, of whom a third will respond to express their interest (N=15,000). Approximately 10% of these will be potentially eligible (N=1500) and sent the full trial information pack, along with an additional 390 people identified by the general practice consultation method. After telephone eligibility screening, 696 people (37%) will be eligible and consent to participate. We will phase in recruitment over the first 4 months (with recruitment targets of 16, 22 and 28 participants respectively, then 30 per month). We anticipate identifying the required 1,900 potentially eligible patients from across up to 60 general practices (denominator adult population 420,000). This estimate is based on the three participant identification and recruitment methods used in our pilot and feasibility trial. However, the pool of potential participants will be increased by using

the additional recruitment methods (screening of physiotherapy/podiatry referrals and self-referral following community awareness-raising).

8.3 Statistical analysis plan

A comprehensive analysis plan will be developed and describe all trial analyses. It will be kept as a separate document to this protocol and represents the *a priori* analysis plan. It will be written using standard operating procedures for Keele CTU and an approved version 1.0 will be signed off by the TSC and DMC committees prior to the start of trial recruitment. Consequently, only a brief outline of the analysis plan is below.

8.3.1 Summary of baseline data and flow of patients

A CONSORT flow diagram, together with appropriate adaptive trial design extensions, will be produced to document the flow of participants through the trial and will include reasons for withdrawal if given. Serious or trial-related adverse events and protocol violations will be reported throughout the trial by treatment arm. Descriptive statistics will be used to describe the key baseline characteristics of participants included at each stage of recruitment and follow-up. The DMC will receive these data and serious or trial-related adverse events and protocol violations disaggregated by trial arm, prepared by the unblinded trial statistician.

8.3.2 Primary outcome analysis and decisions about adaptations to the trial

The main between-group evaluation for the primary outcome will be undertaken using linear mixed modelling adjusted for geographical region treatment site, identification method, age, sex at birth and baseline pain score, by ITT, analysing participants according to their randomised allocation. At stage 1, 87 participants will be recruited to each arm: if a clinician-supported intervention arm is shown to be inferior to control it will be dropped for futility i.e. if it is below the futility bound of t=0.000 (if all clinician-supported interventions are inferior [below the futility bound] the trial will be stopped); if one or more clinician-supported interventions are shown to be superior to control the trial will be stopped for efficacy (i.e. standardised mean group comparison test statistic is above 2.352 for at least one of the intervention arms (versus control), then the trial will be stopped and the arm with the highest test statistics will be recommended). If the trial continues, an additional 87 patients will be recruited to each remaining clinician-supported intervention arm and control. At the end of stage 2, the focus of the pain NRS evaluation will be on superiority testing against an efficacy boundary of t=2.218, and ensuring an overall one-sided 5% type 1 error; hence, if the standardised treatment effect statistic is above 2.218, then the corresponding treatment arm will be recommended. The testing boundaries are constructed under a generalised Dunnett testing procedure using a triangular efficacy boundary and binding futility boundary (i.e. if a test statistic for the corresponding treatment is crossing lower futility boundary, the treatment must be dropped at the interim analysis). If the test statistic for at least one of the treatments crosses the upper efficacy boundary at the time of the interim analysis, the trial will be stopped earlier. If the test statistics for more than one treatment cross the upper efficacy boundary (either at the time of interim or final analyses), all corresponding treatments will be concluded to be superior to the control. Interim and final analysis of the primary pain outcome will be carried out blind to treatment arm allocation and independently by two statisticians to support scientific integrity. As well as the primary endpoint evaluation of the 6-12 week between-group difference in pain severity, we will also evaluate as secondary analyses the between-group differences at individual time-points between weeks 1 to 12 through inclusion of a time x group interaction term. Follow-up pain NRS trajectories will be presented graphically to show the comparative summary data between treatment arms. Further to the primary endpoint (average 6-12 weeks pain NRS), sensitivity analysis of the final primary outcome comparison will include a comparison of complier average causal effect (to ascertain an unbiased estimate of explanatory treatment effect) and analysis of therapist random effects.

8.3.3 Secondary outcome analysis

Between-group evaluation of secondary outcomes will only be undertaken at the end of the trial. Analysis will be through linear and generalised mixed models with aligned link function appropriate to the outcome data being analysed. Analysis will take into account patient-level clustering (repeated measures data) through patient-level random effect and the following fixed effects: geographical region, identification method, age, sex at birth, baseline pain score and corresponding baseline value (as appropriate). Reporting of numbers analysed at each follow-up stage will be given and summary descriptive statistics will be presented per treatment arm that align to the type of data e.g. mean (SD) and/or median (interquartile range) for numeric outcomes; frequency counts and percent per category per treatment arm for categorical outcome variables.

For all primary and secondary analyses, the main between-group estimates will be given along with 95% two-sided confidence intervals and p-values.

8.3.4 Subgroup analyses

Exploratory subgroup analyses of the primary pain outcome will also be carried out (including baseline symptom duration as a potential modifier of treatment effect); the full and exact specification of subgroup variables will be decided upon discussion with the TSC. These will be agreed *a priori* with the TSC and included in the Statistical Analysis Plan. In each case, subgroup variables will be represented in categorical form and exploratory statistical modelling of the subgroup effect will be through statistical interaction of the dummy treatment variable x subgroup variable.

8.3.5 Adjusted analysis

Analyses will take into account patient-level clustering (repeated measures data) through patient-level random effect and the following fixed effects: geographical region, identification method, age, sex at birth, baseline pain score and corresponding baseline value (as appropriate).

8.3.6 Interim analysis and criteria for the premature termination of the trial

At stage 1, 87 participants will be recruited to each arm: if a clinician-supported intervention arm is shown to be inferior to control it will be dropped for futility i.e. if it is below the futility bound of t=0.000 (if all clinician-supported interventions are inferior [below the futility bound] the trial will be stopped); if one or more clinician-supported interventions are shown to be superior to control the trial will be stopped for efficacy (i.e. standardised mean group comparison test statistic is above 2.352 for at least one of the intervention arms (versus control), and the arm with the highest test statistics will be recommended). Any potential requirement to stop any trial arms for safety reasons will be closely monitored by the Data Monitoring Committee.

8.3.7 Participant population

All randomised participants will be analysed according to an ITT approach (analysed as randomised). For any outcome, participants will be included in an analysis if baseline and at least one follow-up measurement is available for that outcome.

8.3.8 Procedure(s) to account for missing or spurious data

The linear and generalised mixed models specified in sections 8.3.3 and 8.3.4 analyse all available data and assume that missingness is 'missing at random' (MAR) rather than 'missing completely at random' (MCAR). Patterns of missing primary outcome data will be explored to check the possibility of the data being 'not missing at random' (NMAR) to which a pattern-mixture model may be applied.

8.4 Economic evaluation

Health Economics Outcomes

Resource use information will comprise participant-specific costs for the type of intervention received including sessions attended and any other resources incurred as part of the intervention. PHP-related resource data will be collected for primary care consultations (e.g. GPs, practice nurses, First Contact Practitioner), visits to other professionals (e.g. hospital consultants, physiotherapists), private healthcare, prescribed medications and hospital-based tests and investigations and procedures, non-

drug treatments (e.g. injections) and inpatient stays. All costs will be collected via participant questionnaires at 6 and 12 months. Unit cost data will be derived from nationally represented sources such as the British National Formulary (BNF), the National Schedule for Reference Costs and the Unit Costs of Health and Social Care. Given the nature of the condition, broader costs related to both out of pocket costs (e.g., over the counter medications) and productivity losses (e.g., time-off work related to PHP and reduced work performance (presenteeism) will be collected. Information on time off work, occupation, typical work activities and the nature of their employment (full time or part time) will be requested, and the Single-Item Presenteeism Question from the Work Productivity and Activity Impairment Questionnaire will be used to estimate productivity losses relating to presenteeism. The average wage for each respondent will be identified using UK Standard Occupational Classification coding and annual earnings data for each job type.

Health Economic Analysis

A full health economics analysis plan (HEAP) will be written in parallel with the Statistical Analysis Plan, prior to the analyses. The plan will be agreed by the trial health economists, and other members of the trial and TSC/DMC prior to the analysis. Consequently, a brief outline of the data analysis plan is included in this protocol. An economic analysis will be undertaken alongside the trial to estimate the cost-effectiveness of the interventions in the final analysis and will adhere to the recommendations of the NICE Reference Case and these will be calculated using EQ-5D-5L responses.

The evaluation will take the form of an incremental cost-utility analysis to estimate the cost per quality adjusted life year (QALY) over 12 months follow-up using data from the trial. The base-case analysis will be from an NHS/personal social services (PSS) perspective, with an additional analysis from a wider societal perspective taking into account costs of productivity loss. The outcome of interest for the economic analysis is the quality-adjusted life years (QALYs), and utility data will be estimated using participant responses obtained from the EQ5D-5L questionnaire administered at all follow-up time points using the "area under the curve" approach. The crosswalk value set will be applied to patient responses to obtain utility scores, in line with current NICE recommendations. A cost-consequence analysis will initially be reported, describing all the important results relating to costs and consequences (across the full range of clinical and cost outcomes). Subsequently, incremental cost-utility analysis will be undertaken to estimate the incremental cost per QALY gained, adjusting for baseline covariates in line with the primary clinical evaluation. The robustness of the results will be explored using sensitivity analysis.

Final outcomes adjusting for the adaptive design of the trial will be expressed as incremental costeffectiveness ratios (ICERs) using ICER plots and cost-effectiveness acceptability curves (CEACs) to represent the probability of being cost-effective at different willingness to pay thresholds. Given the limited consensus in analysing economic evaluations alongside adaptive multi-arm platform trials,^{66,67} decision modelling will also be undertaken as a means of adjusting for the adaptive nature of the trial. The model will be updated as the trial progresses using the interim trial data, literature review of previous relevant models, and stakeholder consultations. The model will be subject to deterministic sensitivity analyses in the final analysis. Cost-effectiveness planes and cost-effectiveness acceptability curves will be presented to show the probability the intervention is cost-effective at different cost/QALY thresholds. Full details of the model approach will be described in the HEAP.

9 DATA HANDLING

9.1 Data collection tools and source document identification

Self-report questionnaires (online and paper-based), treatment diaries, SMS text-messages and CRFs will form the basis of data collection.

9.2 Data handling and record keeping

Data management will be carried out in accordance with a Data Management Plan, in accordance with Keele University Health and Social Care Research (HSCR) Standard Operating Procedures (SOPs). The trial data will be stored on Keele University storage services within the UK and protected by industry standard security tools. All confidentiality arrangements adhere to relevant data protection regulations and guidelines (General Data Protection Regulation (GDPR), Caldicott, General Medical Council (GMC), Medical Research Council (MRC) UK Policy) and the Chief Investigator and the Data Custodian has responsibility for the use, security and management of all data generated by the study.

Questionnaires completed electronically will be date stamped electronically on receipt at Keele CTU. Paper-based questionnaires will be sent to the Keele CTU data management team in pre-paid envelopes provided to participants. Paper-based questionnaires will be date stamped on receipt at Keele CTU. Questionnaires will then be logged as returned on a management database and the participant's responses entered into the trial database; the database will be tested *a priori* for reliability. The trial statistician will determine coding of questionnaire items, in accordance with standardised coding procedures of Keele CTU, to facilitate data entry. Members of the research team will enter data and cross checks (a minimum of 1 in 10) will be carried out by other team members to ensure reliability and quality assessment of data entry.

The trial is designed so that all participant personal data (e.g. names, addresses, email addresses prior to de-identification) are located on a Keele REDCap database hosted by Amazon Web Services (AWS) secure web servers, restricted to authorised personnel. Furthermore, all data used for analysis will be kept separate from participant personal data to ensure anonymity to meet the necessary standards of the Keele CTU data security policy. Similarly, all hard copy information (e.g. signed consent forms, questionnaires) will be stored securely within the Keele CTU in accordance with Standard Operating Procedures (SOPs).

SMS text-messages will operate using a third party text service provider which adheres to the Data Protection Act 2018 as well as the EC Directive "Private and Electronic Communications Regulations 2003". Communication between Keele CTU's secure server and the text-message service provider is both encrypted and authenticated using a unique ID and password.

All protocol deviations are expected to be reported to Keele CTU as soon as the Investigator has become aware of the event. These will be reported accordingly to Keele University's SOPs.

The Data Management Team based at the Keele CTU will oversee all responsibilities delegated to the CTU for data management and data entered to the trial database.

9.3 Access to Data

Direct access to trial-specific data only will be given to authorised representatives of the Sponsor to permit trial monitoring and audit.

9.4 Data Sharing Agreements

Keele CTU is committed to sharing access to our anonymised research data derived from our population, consultation, clinical and RCT cohorts. Requests for access to the anonymised data from anyone outside of the trial team (e.g. collaboration, joint publication, data sharing requests from publishers) will be reviewed. All information will be held securely and in strict confidence. Each person in the trial will be given a unique trial ID so that data will not have any identifiable information, such as names and addresses. On this basis, these anonymised data will be kept electronically and may be used in other research studies.

9.5 Archiving

At the end of the trial, data will be securely archived in line with the Sponsor's procedures for 10 years after end of the trial declaration and until the sponsor authorises destruction. Archiving will be carried out in accordance with Keele University SOPs.

10 MONITORING & AUDIT

10.1 Trial Management

The TREADON trial is fully supported by Keele CTU, a UKCRC registered CTU. An experienced Trials Manager at Keele CTU will be responsible for day-to-day trial management, supported by a CTU Senior Trials Manager. The Trial Manager will support a Trial Coordinator at each of the collaborating academic sites who will liaise with general practices, physiotherapy and podiatry services, and research networks (CRNs and NHS Research Scotland Primary Care Network). The trial is sponsored by Keele University, and will be conducted in line with applicable Keele University Health and Social Care Research and Keele CTU SOPs. Keele CTU will provide trial services for: eligibility screening, consent, data collection and response management, data entry, follow-up administration and minimal data collection; IT to include database development, health informatics, data management, randomisation, and SMS text-message system; statistical design and analysis.

The Chief Investigator (Roddy) will be responsible for financial management of the trial, supported by Keele's Associate Investigator (Thomas), the Senior Trials Manager, Keele CTU and the School of Medicine Finance and Business Manager. Contracts will be developed and in place between the academic institutions and the NHS sites, to clearly articulate roles and responsibilities.

The Trial will be overseen by the Trial Management Group (TMG) chaired by the Chief Investigator (Roddy) or Keele's Associate Investigator (Thomas) and consist of representatives of all key groups involved in the design, operation and management of the trial. This group will meet monthly and will monitor progress along the planned timetable, discussing any issues as they arise and troubleshooting as required. Remote conferencing (e.g. Microsoft Teams) will be utilised for all TMG meetings, as required, to facilitate participation of co-investigators from all institutions.

The lead NIHR Clinical Research Network (CRN) for the delivery of the trial will be West Midlands (WM), with whom we have an excellent relationship. Their remit will involve the identification of sites and associated activities involved in site set-up and the recruitment of eligible participants. York/Humber CRN and the NHS Research Scotland Primary Care Network will work with the clinical teams within their local participating NHS services and local R&D teams. Our experience of working with other networks, includes receiving valuable advice and feedback on the feasibility of planned recruitment methods, and on efficient ways of securing Network infrastructure and service support to secure timely delivery of research. In addition, we engage with resources of West Midlands CRN Physiotherapy Research Facilitators who also provide targeted support for partnering NHS services, site set up and participating physiotherapists and podiatrists. This collaborative approach has been successful in a number of previous trials supported by Keele CTU (e.g. HTA-funded SCOPiC trial⁶⁸ and PROP OA trial⁵⁹). We also support the NIHR CRN West Midlands to achieve 'time to target' and 'first patient in' targets contributing to the highest accrual of NHS patients involved in research nationally. CRN representatives will attend relevant TMG meetings.

An independent TSC that includes two patient representatives, and a DMC will be established and provide independent oversight of the trial. The TSC and DMC will meet initially at the start of the trial to approve the trial protocol and subsequently at regular intervals as agreed by the TSC/DMC.

10.1.1 Trial Steering Committee (TSC) members

See under Key Trial Contacts.

10.1.2 Data Monitoring Committee (DMC) members

See under Key Trial Contacts. The DMC may request to review overall unblinded safety data to determine patterns and trends of events, or to identify safety issues, which would not be apparent on an individual case basis.

10.2 Monitoring arrangements

A trial Risk Assessment will inform a Trial Monitoring Plan. The Keele CTU data management team will perform data quality checks of CRF data. Data queries will be entered to a log which will be sent to the Trial Manager, who will work with each site to resolve data queries in a timely fashion and provide further training as required. This, along with safety reports, will inform a risk-based approach towards assessing a need for any onsite monitoring visits. Trial monitoring reports will be reviewed by the TMG, DMC and TSC.

10.3 Auditing procedures

Keele CTU will conduct internal audits across their research studies in line with the Keele CTU Quality Management System, following a risk-proportionate approach as appropriate.

10.4 Safety Reporting

Collaborating centres should record events or concerns about the safety of subjects that arise as a result of the trial, even if these events or concerns do not meet the definition of a Serious Adverse Event requiring notification to the regulatory authorities.

Adverse Events

The occurrence of adverse events considered to be related to the trial interventions for each intervention will be monitored and assessed using case report forms, contact with the trial team, physiotherapist/podiatrist report, and follow-up questionnaire. Expected adverse events from prefabricated orthoses include: blisters, skin irritation and transient lower limb discomfort during the initial 'wear-in' process. An expected adverse event from unaccustomed exercise and physical activity is temporary, mild muscle soreness. Physiotherapists and Podiatrists delivering the interventions will advise participants about how to manage such symptoms.

Serious Adverse Events (SAE)

An SAE is an untoward event that:

- (a) results in death;
- (b) is life threatening;
- (c) requires hospitalisation or prolongation of existing hospitalisation;
- (d) results in persistent or significant disability/incapacity;
- (e) consists of a congenital anomaly/birth defect; or
- (f) is otherwise considered medically significant by the investigator.

An SAE occurring to a research participant must be reported to the Research Ethics Committee (REC), and to the Sponsor where, in the opinion of the Chief Investigator, the event was *related*, i.e. it resulted from administration of any of the research procedures, and was *unexpected*, i.e. the type of event is not an expected occurrence as a result of the intervention provided. There are no expected SAEs in this trial.

All SAEs considered to be potentially related to the trial procedures and occurring from the point when participants are registered on the trial will be notified to Keele CTU via telephone within 24 hours of the research staff at the site becoming aware of the event. Keele CTU will then provide the appropriate CRF and inform the CI. The CRF must be completed and returned to Keele CTU (via

secure transfer) within one week of receipt, and ideally within 24 hours. Clinicians will be asked to assess whether they considered the SAE was due to the trial procedures. Any follow-up information should be sent to Keele CTU as it is available. Events will be followed up until the event has been resolved or a final outcome has been reached. GPs will be asked in the letter that informs them of their patient's participation in the trial to report to Keele CTU any SAEs that they judge to be potentially related to TREADON trial procedures.

All SAEs either confirmed or suspected to be related to the trial procedures will be reviewed by the DMC and reported to the TSC.

Responsibilities for safety reporting

Investigator (PI) at site

- Using clinical judgement in assigning seriousness, causality.
- Ensuring that SAEs are recorded and reported to Keele CTU in line with requirements of the protocol.

Chief Investigator (CI)

- Clinical oversight of the safety of patients participating in the trial, including an ongoing review of the risk/benefit.
- Assessing seriousness and causality where it has not been possible to obtain local clinical assessment.
- Review of all related SAEs as detailed in the trial monitoring plan.

Keele CTU

- Central data collection and verification of related SAEs.
- Ensuring related SAEs are reported to the TSC and DMC.
- Preparing annual reports to the REC in collaboration with appropriate members of the TMG, and reports required by the sponsor.

10.5 Death notification form

All deaths occurring during the 12 week intervention period must be recorded on the Notification of Participant Death CRF. The trial team will respond to this upon notification, for example if death occurrence is reported by the treating physiotherapist/podiatrist or by a participant's family member contacting Keele CTU.

10.6 Trial timeline

The trial will be delivered over 51 months between January 2022 and March 2026.

10.7 Internal pilot phase

Our pilot and feasibility trial demonstrated feasibility of participant identification and recruitment, intervention fidelity, adherence, and retention at general practices and physiotherapy and podiatry services in the West Midlands.³³ We will undertake an internal pilot phase to test participant identification and recruitment, intervention fidelity, adherence to orthoses and exercises, and response to SMS text-messages across all geographical regions in England and Scotland.

Objectives: Specific objectives of the internal pilot phase of the trial are to:

- 1. check the numbers of recruited participants overall, per month and per region per month
- 2. explore intervention fidelity and participant adherence to exercises and orthoses
- 3. determine response to SMS text-messages.

<u>Methods</u>: The internal pilot will last for 6 months, commencing from the start of recruitment (month 10 of the trial timeline). All regions will participate in the internal pilot. Data collection methods will be as described above.

Outcomes: Outcomes of interest for the internal pilot include:

- 1. numbers of adults with plantar heel pain recruited (overall; per month; per region per month)
- 2. intervention fidelity measured by percentage of participants who received the allocated intervention
- 3. self-reported adherence to orthoses and exercises measured by the percentage reporting performing exercises and/or wearing orthoses as advised
- 4. response to SMS text-messages collecting primary outcome pain NRS data during weeks 6-12.

Sample size: Sixty participants are required across the three clinician-supported intervention arms to be able to estimate the proportion of patients adhering to exercises/orthoses with at least 80% confidence (lower limit 1-sided alpha 0.2) with a 5% margin of error, assuming adherence of 80%.

Progression criteria: A success criteria traffic-light system relating to the internal pilot trial objectives will be used to inform whether we stop at the stage of the internal pilot phase, proceed to the main trial (including the planned interim analyses) or proceed but with protocol amendments (Table 4).

	Do not proceed to	Proceed to main trial with protocol	Proceed to main trial
	main trial	amendments	
Participant recruitment:			
Overall by month 6 of internal pilot recruitment	<78	79-155	≥156
	(<50%)	(50-99%)	(≥100%)
Number recruited per month*	<15	15-29	≥30
Number recruited per region per month*	<5	5-9	≥10
Intervention fidelity:			
Received allocated treatment	<80%	81-99%	100%
Adherence to intervention:			
Performing exercises/wearing orthoses as advised	<60%	61-79%	≥80%
Follow-up:			
6-12 week SMS text-message response	<50%	51-69%	70%

Table 4. Internal pilot progression criteria

*in months 4-6 of internal pilot allow for phased in recruitment over months 1-4.

11 ETHICAL AND TRIAL ADMINISTRATION

This clinical trial has been designed, and will be run, in accordance with the Principles of Good Clinical Practice. The lead Chief Investigator (Roddy) will ensure there is clear delegation of responsibilities within the trial team and will be supported by the lead academic representative (Thomas) and lead representatives from collaborating academic institutions (Hendry, Jaki, Keenan, Kigozi, Menz, Foster). The prefabricated orthoses used in the trial are CE-marked medical devices being used for their intended purpose.

We do not anticipate any major ethical concerns with this trial. All participants will receive at least selfmanagement advice which accords with recommendations in a NICE Clinical Knowledge Summary for PHP⁷. In addition, some participants will also be randomised to receive individualised exercises and/or prefabricated orthoses which are already available within NHS clinical care. All interventions are deemed safe, with only minor expected adverse events (for example skin irritation from orthoses, muscle soreness from exercise) seen in our pilot and feasibility trial. Trial safety reporting procedures will ensure any unexpected serious adverse events which are deemed related to the trial will be reported to the Research Ethics Committee, Sponsor, TSC and DMC. The trial requires the recruitment of patients identified within the NHS to an individually randomised trial involving the collection of primary patient-based data. The CTU will operate this activity in accordance with the Data Protection Act 2018, General Data Protection Regulation (GDPR), other relevant regulations and GCP guidelines. The CTU Secure Infrastructure has been independently audited and achieved level one of the Government backed Cyber Essentials Scheme. Keele University as the Sponsor has a quality management system in place containing standard operating procedures which will be adhered to in the conduct of the trial.

Potentially eligible patients will receive information about the trial and will have time to consider this prior to undertaking telephone screening and prior to providing written informed consent. Participants will be assured of confidentiality and participant identification details will not be made available to anyone outside the trial team. GPs, physiotherapists and podiatrists will be informed of their patients' willingness to be part of the trial (depending on the initial source of participant identification). Those who do not consent to be part of the trial or are ineligible will be asked for their consent to use the information already provided for the trial and given advice to consult their GP (or physiotherapist/podiatrist if identified from physiotherapy/podiatry referral screen) if their PHP continues to be troublesome.

All data collected during the course of the trial will be handled and stored in line with Keele CTU's Data Security procedures and SOPs, which are in accordance with the Data Protection Act 2018, GDPR, other relevant regulations and GCP guidelines. We will anonymise and archive the data for 10 years on a secure server at Keele University. Lead statistician, Lewis, will be the data custodian. We will make the data accessible to other researchers, in line with Keele CTU procedures.

11.1 Research Ethics Committee (REC) review & reports

Before the start of the trial, approval will be sought from a REC for the trial protocol, informed consent forms and patient self-report questionnaires.

- Substantial amendments that require review by REC will not be implemented until the REC grants a
 favourable opinion for the trial (note that amendments may also need to be reviewed by NHS R&D
 departments before they can be implemented in practice at sites).
- All correspondence with the REC will be retained in the Sponsor Trial Management/local Investigator Site File.
- An annual progress report will be submitted to the REC within 30 days of the anniversary date on which the favourable opinion was given, and annually until the trial is declared ended.
- It is the Chief Investigator's responsibility to produce the annual reports as required.
- The Chief Investigator will notify the REC of the end of the trial.
- If the trial is ended prematurely, the Chief Investigator will notify the REC, including the reasons for the premature termination.
- Within one year after the end of the trial, the Chief Investigator will submit a final report with the results, including any publications/abstracts to the REC.

11.2 Peer review

The trial has been funded by the NIHR HTA Programme through open competition and hence has undergone external peer review by appropriate patient and healthcare professional representatives.

11.3 Public and Patient Involvement

We have adopted the approach advocated by INVOLVE (2012) and PPIE has been central to the development of the research question and the TREADON pilot and feasibility trial design.

TREADON pilot and feasibility trial

Prior to the pilot and feasibility trial, we held an advisory workshop, which included four patients with plantar heel pain and 15 physiotherapists/podiatrists. Patients discussed their experiences of PHP and their concerns about the need for effective treatment options that would give good symptom relief and rapidly lead to pain-free walking. These priorities form the basis of the agreed research question. Clinicians identified prescribed exercise and foot orthoses as interventions likely to meet the objective of pain-free walking and achieve good long-term clinical outcomes.

Two further workshops included the patient representatives from the advisory group and members from the Primary Care Centre Versus Arthritis (then Arthritis Research UK Primary Care Centre) PPIE group (who have a range of pain conditions including foot pain). These workshops further developed the interventions, in particular; (i) the choice of foot orthoses, taking account of; the requirement to modify the orthoses to patient needs, the differing requirements of males and females regarding footwear and expectations surrounding adherence to daily orthotic device use, (ii) the exercise programme, including realistic expectations of adherence to regimens, and (iii) the information and advice to be provided within the self-management advice booklet which members were keen to ensure was of high quality without unduly comprising the intended 'usual care' nature of the intervention. The suitability of three different outcome measures selected to reflect the patient stated priority of a rapid response to treatment were also discussed. In particular, patients provided feedback on the principle and practicality of a proposal to collect participants' pain scores weekly via text-message to allow a 'time to response' outcome to be measured which patients identified as important.

Patients/service users with PHP advised us about participant identification methods; design and content of patient-facing documentation including participant information leaflets, questionnaires and text-messages; and design of the exercise, foot orthoses and self-management advice treatments to maximise adherence.

Eleven clinicians and all four patients who participated in the workshop agreed to continuing involvement with the feasibility and pilot trial through membership of an advisory group. Via this advisory group, patients and clinicians have continued to be actively involved with the trial development particularly in relation to; honing the patient journey through the three different methods of recruitment, advice on patient documentation and information leaflets and also the content of questionnaires and text-messages. This involvement contributed to optimising recruitment and retention in the feasibility and pilot trial. Adherence to wearing foot orthoses and to following the exercise recommendations is likely to be key to the outcome of these interventions so we invited further input into this aspect in three areas; (i) refining the clinical protocols to maximise adherence, (ii) discussing methods to maximise completion of adherence questions in the follow up process, (iii) review of the definitions of adherence for purposes of analysis. This helped to underpin fidelity of intervention protocols. Patient input to maximising follow up through mobile phone text messaging (or telephone call where preferred) was also sought.

For trial monitoring purposes, a patient was involved in the Trial Steering Committee (TSC) and discussion regarding use and dissemination of the findings. The Primary Care Centre Versus Arthritis (then Arthritis Research UK Primary Care Centre) has a User Support Worker and PPIE co-ordinator who helped to co-ordinate previous and continued PPIE work. They also provide support such as explaining research methodology, where necessary, to the patient members of both the advisory group and TSC. We also provide users with a glossary of terms used in research and offer access to a training session designed to meet the needs of research users.

We held meetings with the advisory group and PPIE groups to ensure that they were involved with the interpretation of the results of the pilot and feasibility trial and any considerations for the future main randomised controlled trial design. Where appropriate dissemination of findings beyond the traditional academic routes was discussed.

TREADON main trial

We held a further advisory workshop in March 2020 in which members of our Research User Group with experience of plantar heel pain advised on the development of the funding application for this main trial. They told us about the substantial and often prolonged impact of plantar heel pain on their daily activities and work. They reported receiving little treatment from their GP: "I just got told to go away and put my foot up. Now I don't even bother going to the GP". The group shared their concerns about the need for effective treatment options that would give good pain relief. These observations form the basis of our agreed research question.

The group recognised the success of the three methods employed in our pilot and feasibility trial to identify potential participants through general practices. However, they told us that people do not necessarily seek health care for plantar heel pain and, following their advice, we have added community advertising, for example using social media, and self-referral as a route into the trial.

They also advised us about the trial's outcome measures. They viewed pain reduction, but not necessarily pain resolution, as the most important outcome: "I just want to reduce the intensity of pain a bit, that's the main thing because it affects life, what you do, where you go. I can't even go out with the dog". Whilst going to work was considered important, they advised us to examine whether people with plantar heel pain can do their job fully (work presenteeism) as they felt that plantar heel pain sufferers might attend work but have difficulty performing their job. They reviewed the wording of the SMS text-message which will be used to collect pain scores and agreed this was appropriate and acceptable. We also asked them how often the SMS text-message should be sent during the primary outcome phase (first 12 weeks of follow-up). They felt that weekly was best, as participants might forget how they had been if text-messages were sent less often.

The group reviewed the self-management advice booklet and the foot orthoses which they felt were acceptable and appropriate and did not recommend any changes. We discussed strategies to improve adherence to exercise. The group felt that communication about the importance of adherence was the key to optimising adherence and as a result we will modify the training programme to stress the importance of communication and the therapeutic alliance between participants and therapists to improve adherence (as well as other measures outlined in the research plan).

Lay co-investigator Brammar advised on the Plain English Summary.

11.4 Regulatory Compliance

The trial will be conducted in accordance with the principles of Good Clinical Practice (GCP) in research studies and the UK Policy Framework for Health and Social Care Research. Keele CTU have a quality management system in place containing standard operating procedures which will be adhered to in the conduct of the trial. Studies run by Keele CTU may be subject to an audit by Keele University as the Sponsor.

11.5 Protocol compliance

Non-compliance may be identified through any trial activity but in particular through the use of central monitoring procedures such as consent form review or data management, CRF review, site visits and self-reporting by the trial site or participants. Deviations from protocols and GCP may occur in research studies. The majority of these instances are technical non-compliances that do not result in harm to the trial participants, do not compromise data integrity, or significantly affect the scientific value of the reported results of the trial. These technical deviations will be documented, and appropriate corrective and preventative actions will be taken by the research team with responsibility being taken by the CI and where needed with agreement from the TSC.

11.6 Notification of Serious Breaches to GCP and/or the trial protocol

Participating sites are expected to notify Keele CTU as soon as they become aware of a serious breach. A "serious breach" is a breach which is likely to affect to a significant degree:

- The safety or physical or mental integrity of the participants of the trial; or
- The scientific value of the trial.

These will be reported accordingly to Keele University's SOPs.

11.7 Data protection and patient confidentiality

All information collected during the course of the trial will be kept strictly confidential. Information will be held securely on paper and managed electronically by Keele University through Keele CTU. Keele CTU complies with data protection regulations:

- Appropriate storage, restricted access to disposal arrangements for participants personal and clinical details
- Consent from participants for access to their healthcare records by responsible individuals from the research staff or regulatory authorities, where it is relevant to trial participation
- Consent from participants for the data collected from the trial to be used to evaluate safety and develop new research
- All data collection forms that are transferred to and from Keele CTU will be coded with a trial number and will include up to three further participant identifiers: the participant's sex at birth, initials and date of birth
- Where anonymisation (and/or appropriate pseudo-anonymised) of documentation is required, participating centres are responsible for ensuring only the instructed identifiers are present before sending to Keele CTU.

All data will be housed in the CTU infrastructure, which is a secure virtual network requiring two factor authentication in order to access the data stored within. Roles and permissions are applied to users within the network as well as within an application to restrict what data a user can access and operations they can perform. All research staff/CTU operational staff involved in this trial adhere to robust data security procedures and have explicit duties of confidentiality. These practices are written into their employment contracts and are equivalent to the duty placed on NHS staff.

If a participant withdraws consent from further trial intervention and/or further collection of data their data will remain on file and will be included in the final trial analysis.

11.8 Financial and other competing interests for the Chief Investigator, PIs at each site and committee members for the overall trial management

Edward Roddy, Martin Thomas, Gordon Hendry and Anne-Marie Keenan have no financial or other competing interests to declare. The members of the TSC and DMC also have no financial or other competing interests to declare.

11.9 Indemnity

The trial is sponsored by Keele University and therefore Keele University will be liable for negligent harm caused by the design of the trial.

The NHS has a duty of care to patients treated, whether or not the patient is taking part in a trial, and the NHS organisation remains liable for clinical negligence and other negligent harm to patients under this duty of care.

Agreements between the Sponsor and participating NHS organisations detailing trial conduct and the responsibilities to be honoured by each party will be fully executed before the trial can start at the local NHS Trust/Health Board.

11.10 Amendments

The detailed protocol will be updated in response to approved amendments, as required. All amendments will be reviewed and authorised by the Sponsor and submitted for review or information to REC/HRA, as appropriate.

11.11 Post trial care

Participants allocated to intervention arms including prefabricated orthoses will be allowed to keep their orthoses after the end of the trial intervention period. The assessments and treatments by the physiotherapist or podiatrist will stop once the participant has completed the 12-week intervention period. Participants will not be offered further healthcare as part of their trial involvement once they have been discharged by their treating physiotherapist/podiatrist following their episode of care within the trial. The participants randomised to either exercise or foot orthoses will have the opportunity to continue with the advice and exercises given and/or continue to use the orthoses.

11.12 Access to the final trial dataset

At the end of the trial, archiving of essential trial documents at Keele University will be authorised by the Sponsor following submission of end of trial reports, which will be for ten years after the end of the trial. Destruction of essential documents will be carried out in accordance with Keele University SOPs.

All data will be held by Keele CTU and will be archived in the designated Keele CTU archive facility. Arrangements for the destruction of all confidential data will be carried out in accordance with Keele University SOPs.

Any subsequent requests for access to the data from anyone outside of Keele CTU (e.g. collaboration, joint publication, data sharing requests from publishers) will be carried out in accordance with Keele University SOPs.

12 DISSEMINATION POLICY

12.1 Dissemination policy

The findings of this randomised trial will be of direct relevance to patients with PHP and the GPs, physiotherapists and podiatrists in primary care who care for them.

Key outputs will be:

- new evidence regarding the clinical and cost-effectiveness of self-management advice (SMA), individualised exercises and prefabricated orthoses for the management of PHP in primary care which will directly inform treatment decisions
- a high-quality SMA booklet that can be used to provide information to people with PHP alongside other treatments
- training materials and treatment protocols to enhance the skills of physiotherapists and podiatrists to prescribe individualised exercises and prefabricated orthoses in the treatment of people with PHP
- dissemination of results as they become available via our website
- open-access publications in high quality peer-reviewed journals
- presentations at conferences attended by the most relevant stakeholders including GPs (e.g. Royal College of General Practitioners (RCGP) conference, Society for Academic Primary Care), podiatry (College of Podiatrists), physiotherapists (Physiotherapy UK) and rheumatology (e.g. British Society for Rheumatology (BSR))

• the team's strong links with patient, professional, third sector and NHS organisations will allow the evidence generated to influence national and international clinical guidelines

• working with our Research User Group (RUG), we will disseminate our findings in ways that will allow patients to better understand the effectiveness of pre-fabricated orthoses and individualised exercises for PHP.

Informing and engaging patients/service users, carers, NHS, Social Care organisations and the wider population

We will make all outputs from this work widely and freely available to all stakeholders, in ways that are easy to access at no cost. In addition to publication in open-access journals, we will use our website and social media, via our dedicated and widely followed Twitter and Facebook feeds, to support dissemination of our findings. Our lay co-investigator and RUG will advise on how to translate the findings into easily understandable messages and on how best to disseminate the results to the wider public. We have strong links with Versus Arthritis (formerly Arthritis Research UK), which will allow us to support patient and clinician education initiatives and promote best practice in the management of musculoskeletal disorders. The team has strong links with patients, NHS organisations and into the key professional networks needed to engage patient, clinical and professional partners, facilitating dissemination to all stakeholders and allowing us to influence national and international clinical guidelines. The trial investigators include members of national (e.g. NICE) guideline groups and work closely with the RCGP, College of Podiatry, Chartered Society of Physiotherapy, Primary Care Rheumatology and Musculoskeletal Medicine Society, and British Society for Rheumatology. The TMG also includes international team members who will facilitate dissemination internationally.

Informing the health and care system and society as a whole

Informed by NIHR 'Push the Pace' guidance, dissemination of the outputs arising from this research will be supported by the Impact Accelerator Unit (IAU) within Keele University's School of Medicine. The IAU, led by Dziedzic, NICE Fellow and NIHR Knowledge Mobilisation Fellow and Senior Investigator, has a strong multidisciplinary team of experts in implementation and knowledge mobilisation and works with local, regional, national and international bodies to support proactive translation of research into clinical practice. Previous projects include STarT Back and JIGSAW-E implementation (https://www.keele.ac.uk/pcsc/research/impactacceleratorunit/projects/) which have been adopted into national and international guidelines and widely taken up into clinical practice. We are committed to ensuring our research influences clinical practice and guidelines. With the support of the IAU, we will seek to optimise dissemination to facilitate translation into clinical practice. Existing strong NHS partnerships will ensure that the research findings are made readily available to general practitioners, physiotherapists, podiatrists, rheumatologists and Integrated Care Systems (ICSs). We will work closely with the NIHR Applied Research Collaborations (co-investigator Mallen is West Midlands long-term conditions theme lead), Academic Health Science Network (AHSN), RCGP, College of Podiatry, Chartered Society of Physiotherapy and British Society for Rheumatology to support uptake of the findings. We have very strong links, formalised through our NHS Consortium, ICSs (via our former links with CCGs), NHS Trusts (including community Trusts with experience of recruiting to RCTs), Public Health England and the Sustainability and Transformation Partnership. A shared research strategy ensures relevant research findings are 'pulled' into our key partnerships and established as routine practice.

Further funding/support options

Our extensive networks, described above, will allow us to rapidly disseminate findings and maximise impact. The team includes opinion leaders capable of driving dissemination and, by working with the AHSN, IAU, Versus Arthritis and professional bodies (e.g. Royal College of Podiatry, Chartered Society of Physiotherapy) we can draw on additional expert resource to pump prime rapid adoption into clinical and guidelines. Depending on the trial findings, we will apply to the AHSN to support wider roll out and to develop educational initiatives to support implementation. We have an established track record in this area, influencing national and international clinical and commissioning guidance through our research.

Potential barriers to future research, development, adoption and implementation

All background intellectual property (IP) relating to this trial is held by Keele University. All foreground IP will be managed by Keele University. Keele University is involved in building a research framework into the Integrated Care System model, and chair a Staffordshire and Shropshire Health Economy Research Partnership (SSHERPA) to deliver a joint research strategy. Keele University will ensure that all outputs from the proposed trial are clearly identified, disseminated broadly and made freely available to policymakers, the public and health care professionals. Early identification of potential IP is supported by Keele's Partnership Development team who are responsible for identifying and supporting knowledge transfer opportunities, IP protection, dissemination and NHS partnerships. The aim of our joint free and open-access IP policy is to actively facilitate rapid translation of research into clinical practice for the benefit of patients, clinicians and NHS services. We do not anticipate any regulatory hurdles to dissemination or adoption of the research outputs into clinical practice. However, we recognise that achieving behaviour change in clinicians is challenging. The questions we will answer are of direct interest to our wider stakeholder group. Harnessing the combined strengths of our collaborating academic institutions and working with key professional bodies and organisations committed to getting research into practice (e.g. AHSN, NIHR Applied Research Collaborations) will facilitate translation of our findings into changes in clinical practice and minimise barriers to uptake.

Potential impact of research

This work will impact on a broad range of stakeholders including patients, clinicians, and healthcare commissioners. It was co-designed with patients to ensure relevance and to maximise potential future impact. Current recommendations for primary care management of PHP advocate initial treatment with analgesia and advice, with referral to a physiotherapist or podiatrist only recommended if symptoms persist beyond a few months. Patients often experience chronic symptoms and inability to work, yet few are referred to a physiotherapist or podiatrist. However, evidence regarding the effectiveness of SMA, individualised exercises and prefabricated orthoses, alone or in combination, is limited. By addressing this evidence gap, we will help patients and clinicians better understand the effectiveness of these interventions for PHP and allow them to make more informed treatment decisions. Healthcare commissioners will be provided with evidence of the clinical and cost-effectiveness of these interventions on which to base the commissioning of services, increasing the availability of these treatments for people with PHP. At the end of the trial, training materials and treatment protocols will be readily and immediately available and we will be well placed to support scale-up of the training programme to enable physiotherapists and podiatrists to deliver evidence-based treatment for PHP.

Sharing trial progress and results with trial participants

We will share our trial findings with participants via our website, dedicated social media feeds (Facebook, Twitter) and posters in participating general practices and physiotherapy and podiatry services. Our lay co-investigator and RUG will advise us on how best to disseminate the results to participants and the wider public and will help us to translate the findings into easily understandable messages. We will provide a summary to all trial participants describing the trial findings.

12.2 Authorship eligibility guidelines and any intended use of professional writers

Authorship will be available to those who fulfil the International Committee of Medical Journal Editors (ICMJE) criteria (<u>http://www.icmje.org/recommendations/browse/roles-and-responsibilities/defining-the-role-of-authors-and-contributors.html</u>) and in accordance with the requirements and guidance for authors from the NIHR Journals Library (<u>https://www.journalslibrary.nihr.ac.uk/information-for-authors/</u>).

Staff heavily involved in the practicalities of trial operationalisation and delivery, including dedicated study co-ordinators, will be considered for co-authorship of protocol papers on the condition they can contribute to critical revision of drafts, approve the final version, and be accountable for the content. Professional writers will not be used.

13 REFERENCES

- 1. Riel H, Cotchett M, Delahunt E, Rathleff MS, Vicenzino B, Weir A, Landorf KB. Is 'plantar heel pain' a more appropriate term than 'plantar fasciitis'? Time to move on. Br J Sports Med 2017;51(22):1576-1577.
- 2. Thomas MJ, Whittle R, Menz HB, Rathod-Mistry T, Marshall M, Roddy E. . Plantar heel pain in middle-aged and older adults: population prevalence, associations with health status and lifestyle factors, and frequency of healthcare use. BMC Musculoskeletal Disord 2019;20(1):337.
- 3. Tong KB, Furia J. Economic burden of plantar fasciitis treatment in the United States. Am J Orthop 2010;39(5):227-231.
- 4. Irving DB, Cook JL, Young MA, Menz HB. Impact of chronic plantar heel pain on health-related quality of life. J Am Podiatr Med Assoc 2008;98(4):283-289.
- 5. Irving DB, Cook JL, Menz HB. Factors associated with chronic plantar heel pain: a systematic review. J Sci Med Sport 2006;9(1-2):11-22.
- 6. Osborne JWA, Menz HB, Whittaker GA, Landorf KB. Muscle Function and Muscle Size Differences in People With and Without Plantar Heel Pain: A Systematic Review. J Orthop Sports Phys Ther 2019;49(12):925-933.
- 7. NICE Clinical Knowledge Summaries. Plantar Fasciitis. 2015. http://cks.nice.org.uk/plantarfasciitis. [accessed 07 January 2022]
- Hansen L, Krogh TP, Ellingsen T, Bolvig L, Fredberg. Long-Term Prognosis of Plantar Fasciitis: A 5- to 15-Year Follow-up Study of 174 Patients With Ultrasound Examination. Orthop J Sports Med 2018;6(3):2325967118757983.
- 9. Pollack A, Britt H. Plantar fasciitis in Australian general practice. Aust Fam Physician 2015;44(3):90-91.
- 10. Rasenberg N, Bierma-Zeinstra SM, Bindels PJ, van der Lei J, van Middelkoop M. Incidence, prevalence, and management of plantar heel pain: a retrospective cohort study in Dutch primary care. Br J Gen Pract 2019;69(688):e801-808.
- 11. Roddy E, Myers M, Thomas MJ, Marshall M, D'Cruz D, Menz HB, Belcher J, Muller S, Peat G. The clinical assessment study of the foot (CASF): study protocol for a prospective observational study of foot pain and foot osteoarthritis in the general population. J Foot Ankle Res 2011;4:22.
- 12. Cotchett M, Rathleff MS, Dilnot M, Landorf KB, Morrissey D, Barton C. Lived experience and attitudes of people with plantar heel pain: a qualitative exploration. J Foot Ankle Res 2020;13(1):12.
- 13. Grieve R, Palmer S. Physiotherapy for plantar fasciitis: a UK-wide survey of current practice. Physiotherapy 2017;103(2):193-200.
- 14. Telfer S, Abbott M, Steultjens M, Rafferty D, Woodburn J. Dose-response effects of customised foot orthoses on lower limb muscle activity and plantar pressures in pronated foot type. Gait Posture 2013;38(3):443-449.
- 15. Mills K, Blanch P, Chapman AR, McPoil TG, Vicenzino B. Foot orthoses and gait: a systematic review and meta-analysis of literature pertaining to potential mechanisms. Br J Sports Med 2009;44(14):1035-1046.
- 16. Hsu YC, Gung YW, Shih SL, Feng CK, Wei SH, Yu CH, Chen CS. Using an optimization approach to design an insole for lowering plantar fascia stress—a finite element study. Ann Biomed Eng 2008;36(8):1345-1352.
- 17. Chen WP, Ju CW, Tang FT. Effects of total contact insoles on the plantar stress redistribution: a finite element analysis. Clin Biomech 2003;18(6):S17-24.

- 18. Chia KK, Suresh S, Kuah A, Ong JLJ, Phua JMT, Seah AL. Comparative trial of the foot pressure patterns between corrective orthotics, formthotics, bone spur pads and flat insoles in patients with chronic plantar fasciitis. Ann Acad Med Singapore 2009;38(10):869-875.
- 19. McPoil TG, Martin RL, Cornwall MW, Wukich DK, Irrgang JJ, Godges JJ. Heel pain-plantar fasciitis: clinical practice guidelines linked to the international classification of function, disability, and health from the orthopaedic section of the American Physical Therapy Association. J Orthop Sports Phys Ther 2008;38(4):A1-18.
- 20. Young CC, Rutherford DS, Niedfeldt MW. Treatment of plantar fasciitis. Am Fam Physician 2001;63(3):467-74.
- 21. Harty J, Soffe K, O'Toole G, Stephens MM. The role of hamstring tightness in plantar fasciitis. Foot Ankle Int 2005;26(12):1089-1092.
- 22. Whittaker GA, Munteanu S, Menz HB, Tan JM, Rabusin CL, Landorf KB. Foot orthoses for plantar heel pain: a systematic review and meta-analysis. Br J Sports Med 2018;52(5):322-328.
- 23. Rasenberg N, Riel H, Rathleff MS, Birema-Zeinstra SMA, van Middelkoop M. Efficacy of foot orthoses for the treatment of plantar heel pain: a systematic review and meta-analysis. Br J Sports Med 2018;52(16):1040-1046.
- 24. Babatunde OO, Legha A, Littlewood C, Chesterson LS, Thomas MJ, Menz HB, van der Windt D, Roddy E. Comparative effectiveness of treatment options for plantar heel pain: a systematic review with network meta-analysis. Br J Sports Med 2019;53(3):182-194.
- 25. Whittaker GA, Munteanu SE, Menz HB, Landorf KB. Should foot orthoses be used for plantar heel pain? Br J Sports Med 2018;52(19):1224-1225.
- 26. Pfeffer G, Bacchetti P, Deland J, Lewis A, Anderson R, Davis W, Alvarez R, Brodsky J, Cooper P, Frey C, Herrick R, Myerson M, Sammarco J, Janecki C, Ross S, Bowman M, Smith R. Comparison of custom and prefabricated orthoses in the initial treatment of proximal plantar fasciitis. Foot Ankle Int 1999;20(4):214-221.
- 27. McClinton SM, Heiderscheit BC, McPoil TG, Flynn TW. Effectiveness of physical therapy treatment in addition to usual podiatry management of plantar heel pain: a randomized clinical trial. BMC Musculoskelet Disord 2019;20(1):630.
- 28. Rasenberg N, Bierma-Zeinstra SMA, Fuit L, Rathleff MS, Dieker A, van Veldhoven P, Bindels PJE, van Middelkoop M. Custom insoles versus sham and GP-led usual care in patients with plantar heel pain: results of the STAP-study a randomised controlled trial. Br J Sports Med 2021;55(5):272-278.
- 29. Menz HB, Cotchett MP, Whittaker GA, Munteanu SE, Landorf KB. Plantar heel pain: should you consult a general practitioner or a podiatrist? Br J Sports Med 2021;55(5):245-246.
- 30. ISRCTN82762757. Shockwave therapy and exercise for plantar heel pain: A pilot study. https://doi.org/10.1186/ISRCTN82762757 [accessed 07 January 2022].
- 31. Franettovich Smith MM, Collins NJ, Mellor R, Grimaldi A, Elliott J, Hoggarth M, Weber Ii KA, Vicenzino B. Foot exercise plus education versus wait and see for the treatment of plantar heel pain (FEET trial): a protocol for a feasibility study. J Foot Ankle Res 2020;13(1):20.
- 32. Ring K, Otter S. Clinical efficacy and cost-effectiveness of bespoke and prefabricated foot orthoses for plantar heel pain: a prospective cohort study. Musculoskeletal Care 2014;12(1):1-10.
- 33. Chesterton LS, Thomas MJ, Hendry G, Chen Y, Goddin D, Halliday N, Lawton SA, Lewis M, Mallen CD, Menz HB, Foster NE, Roddy E. Self-management advice, exercise and foot orthoses for plantar heel pain: the TREADON pilot and feasibility randomised trial. Pilot Feasibility Stud 2021;7(1):92.

- 34. Budiman-Mak E, Conrad KJ, Roach KE.. The Foot Function Index: a measure of foot pain and disability. J Clin Epidemiol 1991;44(6):561-570.
- 35. Garrow AP, Papageorgiou AC, Silman AJ, Thomas E, Jayson MI, Macfarlane GJ. Development and validation of a questionnaire to assess disabling foot pain. Pain 2000;85(1-2):107-113.
- 36. Menz HB, Auhl M, Ristevski S, Fresco N, Munteanu SE. Comparison of the responsiveness of the foot health status questionnaire and the Manchester foot pain and disability index in older people. Health Qual Life Outcomes 2014;12:158.
- 37. Van der Zwaard BC, Terwee CB, Roddy E, Terluin B, van der Horst HE, Elders PJ. Evaluation of the measurement properties of the Manchester foot pain and disability index. BMC Musculoskelet Disord 2014;15:276.
- 38. Nicholas MK. The pain self-efficacy questionnaire: Taking pain into account. Eur J Pain 2007;11(2):153-163.
- 39. Broadbent E, Petrie KJ, Main J, Weinman J. The brief illness perception questionnaire. J Psychosom Res 2006;60(6): 631-637.
- 40. Herdman M, Gudex C, Lloyd A, Janssen MF, Kind P, Parkin D, Bonsel G, Badia X. Development and preliminary testing of the new five-level version of EQ-5D (EQ-5D-5L). Qual Life Res 2011;20(10):1727-1736.
- 41. Kigozi J, Lewis M, Jowett S, Barton P, Coast J.. Construct validity and responsiveness of the single-item presenteeism question in patients with lower back pain for the measurement of presenteeism. Spine 2014;39(5):409-416.
- 42. Versus Arthritis. Plantar Fasciitis Information and Exercise Sheet [accessed 10 January 2022]. https://www.versusarthritis.org/media/21790/plantar-exercise-sheet.pdf
- 43. Versus Arthritis. Foot and Ankle Pain Booklet [accessed 10 January 2022]. https://www.versusarthritis.org/media/23896/foot-and-ankle-pain-information-booklet.pdf
- 44. Morrisey D, Cotchett M, J'Bari AS, Prior T, Griffiths IB, Rathleff MS, Gulle H, Vicenzino B, Barton CJ. Management of plantar heel pain: a best practice guide informed by a systematic review, expert clinical reasoning and patient values. Br J Sports Med 2021;55(19):1106-1118.
- 45. Garber CE, Blissmer B, Deschenes MR, Franklin BA, Lamonte MJ, Lee IM, Nieman DC, Swain DP, American College of Sports Medicine. American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. Med Sci Sports Exerc 2011;43(7):1334-1359.
- 46. Rathleff MS, Mølgaard CM, Fredberg U, Kaalund S, Andersen KB, Jensen TT, Aaskov S, Olesen JL. High-load strength training improves outcomes in patients with plantar fasciitis: a randomised controlled trial with 12-month follow-up. Scand J Med Sci Sports 2015;25(3):e292-300.
- 47. Sweeting D, Parish B, Hooper L, Chester R. The effectiveness of manual stretching in the treatment of plantar heel pain: a systematic review. J Foot Ankle Res 2011;4:19.
- 48. Foster NE, Thomas E, Barlas P, Hill JC, Young J, Mason E, Hay EM. Acupuncture as an adjunct to exercise based physiotherapy for osteoarthritis of the knee: randomised controlled trial. BMJ 2007;335(7617):436.
- 49. Roddy E, Ogollah RO, Oppong R, Zwierska I, Datta P, Hall A, Hay E, Jackson S, Jowett S, Lewis M, Shufflebotham J, Stevenson K, van der Windt DA, Young J, Foster NE. Optimising outcomes of exercise and corticosteroid injection in patients with subacromial pain (impingement) syndrome: a factorial randomised trial. Br J Sports Med 2021;55(5):262-271.

- Redmond AC, Crosbie J, Ouvrier RA. Development and validation of a novel rating system for scoring standing foot posture: the Foot Posture Index. Clin Biomech (Bristol, Avon) 2006;21(1):89-98.
- 51. Munteanu SE, Landorf KB, Menz HB, Cook JL, Pizzari T, Scott LA. Efficacy of customised foot orthoses in the treatment of Achilles tendinopathy: study protocol for a randomised trial. J Foot Ankle Res 2009;2:27.
- 52. Landorf K, Keenan AM, Rushworth RL. Foot orthosis prescription habits of Australian and New Zealand podiatric physicians. J Am Podiatr Med Assoc 2001;91(4):174-183.
- 53. Kinney M, Seider J, Beaty AF, Coughlin K, Dyal M, Clewley D et al. The impact of therapeutic alliance in physical therapy for chronic musculoskeletal pain: A systematic review of the literature. Physiother Theory Pract 2020;36(8):886-898.
- 54. Babatunde F, MacDermid J, MacIntyre N. Characteristics of therapeutic alliance in musculoskeletal physiotherapy and occupational therapy practice: a scoping review of the literature. BMC Health Serv Res 2017;17(1):375.
- 55. Menz HB, Jodran KP, Roddy E, Croft PR. Characteristics of primary care consultations for musculoskeletal foot and ankle problems in the UK. Rheumatology (Oxford) 2010;49(7):1391-1398.
- 56. Garrow AP, Silman AJ, Macfarlane GJ. The Cheshire Foot Pain and Disability Survey: a population survey assessing prevalence and associations. Pain 2004;110(1-2):378-84.
- 57. Chatterton BD, Muller S, Thomas MJ, Menz HB, Rome K, Roddy E. Inter and intra-rater repeatability of the scoring of foot pain drawings. J Foot Ankle Res 2013;6(1):44.
- 58. Foster NE, Bishop A, Bartlam B, Ogollah R, Barlas P, Holden M, Ismail K, Jowett S, Kettle C, Kigozi J, Lewis M, Lloyd A, Waterfield J, Young Y. Evaluating Acupuncture and Standard carE for pregnant women with Back pain (EASE Back): a feasibility study and pilot randomised trial. Health Technology Assessment 2016;20(33):1-236.
- 59. Holden MA, Callaghan M, Felson D, Birrell F, Nicholls E, Jowett S, Kigozi J, McBeth J, Borrelli B, Jinks C, Foster NE, Dziedzic K, Mallen C, Ingram C, Sutton A, Lawton S, Halliday N, Hartshorne L, Williams H, Browell R, Hudson H, Marshall M, Sowden G, Herron D, Asamane E, Peat G. Clinical and cost-effectiveness of bracing in symptomatic knee osteoarthritis management: protocol for a multi-centre, primary care, randomised, parallel-group, superiority trial. BMJ Open 2021;11(3):e048196.
- 60. Hay E, Dziedzic K, Foster N, Peat G, van der Windt D, Bartlam B, Blagojevic-Bucknall M, Edwards J, Healey E, Holden MA, Hughes R, Jinks C, Jordan K, Jowett S, Lewis M, Mallen C, Morden A, Nicholls E, Ong BN, Porcheret M, Wulff J, Kigozi J, Oppong R, Paskins Z, Croft P. Optimal primary care management of clinical osteoarthritis and joint pain in older people: a mixed-methods programme of systematic reviews, observational and qualitative studies, and randomised controlled trials. Southampton (UK): NIHR Journals Library; 2018.
- 61. Axén I, Bodin L, Bergström G, Halasz L, Lange F, Lövgren PW, Rosenbaum A, Leboeuf-Yde, Jensen I. The use of weekly text messaging over 6 months was a feasible method for monitoring the clinical course of low back pain in patients seeking chiropractice care. J Clin Epidemiol. 2012;65(4):454-561.
- 62. Olsen MF, Bjerre E, Hansen MD, Hilden J, Landler NE, Tendal B, Hróbjartsson A. Pain relief that matters to patients: systematic review of empirical studies assessing the minimum clinically important difference in acute pain. BMC Med 2017;15(1):35.
- 63. Cohen J. Statistical Power Analysis for the Behavioral Sciences. 1988. New York, NY: Routledge.

- 64. Vickers AJ. How many repeated measures in repeated measures designs? Statistical issues for comparative trials. BMC Med Res Methodol 2003;27;3:22.
- 65. Magirr D, Jaki T, Whitehead J. A generalized Dunnett test for multi-arm multi-stage clinical studies with treatment selection. Biometrika 2012;99(2):494-501.
- 66. Flight L, Julious S, Brennan A, Todd S, Hind D. How can health economics be used in the design and analysis of adaptive clinical trials? A qualitative analysis. Trials 2020;21(1):252.
- 67. Flight L, Arshad F, Barnsley R, Patel K, Julious S, Brennan A, Todd S. A review of clinical trials with an adaptive design and health economic analysis. Value Health 2019;22(4):391-398.
- 68. Konstantinou K, Lewis M, Dunn KM, Ogollah R, Artus M, Hill JC, Hughes G, Robinson M, Saunders B, Bartlam B, Kigozi J, Jowett S, Mallen CD, Hay EM, van der Windt DA, Foster NE. Stratified care versus usual care for management of patients presenting with sciatica in primary care (SCOPiC): a randomised controlled trial. Lancet Rheumatol 2020;2(7):e401-e411.
- 69. Hill JC, Garvin S, Chen Y, Cooper V, Wathall S, Saunders B, Lewis M, Protheroe J, Chudyk A, Dunn KM, Hay E, van der Windt D, Mallen C, Foster NE. Stratified primary care versus nonstratified care for musculoskeletal pain: findings from the STarT MSK feasibility and pilot cluster randomized controlled trial. BMC Fam Pract 2020;21(1):30.
- 70. Menz HB, Auhl M, Tan JM, Levinger P, Roddy E, Munteanu S. Effectiveness of foot orthoses versus rocker-sole footwear for first metatarsophalangeal joint osteoarthritis: randomised trial. Arthritis Care Res 2016;68(5):581-589.
- 71. Chesterton LS Blagojevic-Bucknall M, Burton C, Dziedzic KS, Davenport G, Jowett SM, Myers HL, Oppong R, Rathod-Mistry T, van der Windt D, Hay EM, Roddy E. The clinical and cost effectiveness of corticosteroid injection compared with night splints for carpal tunnel syndrome: An open label; parallel group randomised clinical trial (The INSTINCTS trial). Lancet 2018;392(10156):1423-1433.
- 72. Paskins Z, Bromley K, Lewis M, Hughes G, Hughes E, Hennings S, Cherrington A, Hall A, Holden MA, Stevenson K, Menon A, Roberts P, Peat G, Jinks C, Kigozi J, Oppong R, Foster NE, Mallen CD, Roddy E. Clinical effectiveness of one ultrasound guided intra-articular corticosteroid and local anaesthetic injection in addition to advice and education for hip osteoarthritis (HIT trial): single blind, parallel group, three arm, randomised controlled trial. BMJ 2022;377:e068446.
- 73. Pushpakom S, Kolamunnage-Dona R, Taylor C, Foster T, Spowart C, García-Finana M, Kemp GJ, Jaki T, Khoo S, Williamson P, Pirmohamed M, TAILoR Study Group. TAILoR (TelmisArtan and InsuLin Resistance in Human Immunodeficiency Virus [HIV]): An Adaptive-design, Dose-ranging Phase IIb Randomized Trial of Telmisartan for the Reduction of Insulin Resistance in HIV-positive Individuals on Combination Antiretroviral Therapy. Clin Infect Dis 2020;70(10):2062-2072.
- 74. RECOVERY Collaborative Group; Horby P, Lim WS, Emberson JR, Mafham M, Bell JL, Linsell L, Staplin N, Brightling C, Ustianowski A, Elmahi E, Prudon B, Green C, Felton T, Chadwick D, Rege K, Fegan C, Chappell LC, Faust SN, Jaki T, Jeffery K, Montgomery A, Rowan K, Juszczak E, Baillie JK, Haynes R, Landray MJ. Dexamethasone in Hospitalized Patients with COVID-19. N Eng J Med 2021;384(8):693-704.

14. APPENDICIES

14.1 Appendix 1 - Trial management / responsibilities

We are a multi-disciplinary team with representation from general practice, rheumatology, physiotherapy, podiatry, biostatistics, health economics, adaptive trial design, and PPIE. We have internationally-recognised expertise in PHP, exercise interventions, foot orthoses, and RCT design and delivery. We have an experienced lay co-investigator with strong institutional PPIE support. Keele CTU has expertise in delivering large RCTs in primary care.

The trial requires considerable understanding of PHP, exercise interventions, and orthoses in the context of primary care and NHS physiotherapy and podiatry services and the co-investigator team and collaborators have this relevant expertise. Roddy, Keenan, Hendry, and Menz are international leaders in the fields of musculoskeletal foot problems including PHP. We have extensive clinical and research expertise in exercise interventions (Foster, Holden, Thomas) and foot orthoses (Keenan, Hendry, Menz) for musculoskeletal problems, and extensive expertise in primary care and musculoskeletal health (Mallen, Foster, Roddy, Keenan, Menz).

The team also has extensive experience and expertise in trial methodology (Lewis, Foster, Menz, Roddy, Keenan, Mallen), adaptive trial design (Jaki, Mozgunov, Burnett), biostatistics (Lewis), and health economics (Kigozi). Our PPIE co-investigator (Brammar) will ensure the patient perspective is heard throughout the whole research process.

We have a strong track record of successful delivery of trials with high recruitment and follow-up rates, including in musculoskeletal and foot problems in primary care and trials using adaptive designs.^{48,49,68-74} Our research is published in the top medical journals (e.g. Lancet, BMJ). Keele CTU has extensive experience in delivering trials in this field and has established NHS and University links. Recent trials in this field supported by Keele CTU recruited to target demonstrating our ability to deliver this trial in successful partnership with NHS clinicians and services and CRNs.

Roddy (Chief Investigator (CI)) will have overall responsibility for the trial. Thomas has completed a NIHR Development and Skills Enhancement Award in Clinical Trials (2020-2021), a stated goal of which was to be Academic Principal Investigator of the TREADON trial, if funded, under the supervision of Roddy. He will work closely with Keele CTU and the PIs for Glasgow (Hendry) and Leeds (Keenan) to provide day-to-day trial oversight. Foster, Hendry, Holden, Keenan, Thomas, and Menz will design and deliver the training programme for trial physiotherapists and podiatrists, building on the programme used in our pilot and feasibility trial.

Lewis (statistical lead) will be responsible for generating the statistical analysis plan, TSC/DMC reports, conducting data analysis, and supervising the junior statistician. Mozgunov, Jaki and Burnett will provide expertise in adaptive trial design and analysis and will support Lewis and junior statistician to undertake the interim and final analyses. Kigozi will undertake the economic evaluation. Brammar, our PPIE representative, will ensure the patient view actively informs the whole of the trial, supported by Thomas (PPIE lead).

Contribution of collaborators: Dr Thomas Burnett (University of Bath) will provide further expertise in adaptive trial design and analysis and will support co-investigator Mozgunov and Jaki. Professor Hylton Menz (La Trobe University, Melbourne, Australia) is an international leader in podiatry and will contribute considerable expertise and experience in plantar heel pain, randomised trials and foot orthoses.